# Literature Review on Enhancing Integration of Disaster Risk and Climate Change Adaptation in Irish Emergency Planning 13/07/2020

Greene, S., Medway, P., Cubie, D., Le Tissier, M.





This project is funded under the EPA Research Programme 2014-2020. The EPA Research Programme is a Government of Ireland initiative funded by the Department of Communications, Climate Action and Environment. It is administered by the Environmental Protection Agency, which has the statutory function of co-ordinating and promoting environmental research



# **Table of Contents**

Table of Acronyms	i
Key Definitions	iii
Introduction & Background	1
1. CCA & DRM – Making the Connections	4
1.1 Historical alignment and disparities between CCA and DRM	4
1.2 Disaster Risk Management (DRM) and Disaster Risk Reduction (DRR)	4
2. CCA & DRM in Ireland – An Overview	6
2.1 An Overview of Relevant Legal & Policy Frameworks in Ireland	6
2.2 Disaster Risk Management	7
National Context	7
Response and Risk Management Authorities in Ireland	8
Local Level Coordination	8
European Context	9
International Context	9
Ireland's Integration of SFDRR in the SEM Framework	10
2.3 Climate Change Adaptation	11
National Context	12
Local Level Coordination	13
European Context	15
International Context	15
2.4 Ireland and the UN 2030 Agenda for Sustainable Development	16
2.5 Mapping of Key Actors and Stakeholders in Ireland	18
2.6 Resilience and Vulnerability in CCA & DRM in an International Context	22
2.7 Resilience and Vulnerability in CCA & DRM in Irish Frameworks	24
3. Challenges, Gaps and Opportunities in Ireland	25
3.1 Examples of Good Practices	25
Flooding	25
The CFRAM Programme	27
Nature-Based Flood Relief Scheme	27
Infrastructure and Transport	28
Health	29
3.2 Challenges and Barriers in Integration	30

Institutional Arrangements	31
Unequal Funding	32
Misaligned Time Scales	33
Knowledge Gap	33
Language Barrier	35
Strengthening Participation	36
4. CCA & DRM – Bridging the Gaps	37
4.1 Risk and Adaptation Financing	37
4.2 Cross-sectoral Communication, Coordination & Collaboration	38
Local Level Integration	39
4.3 Envisioning a Harmonised Language for CCA & DRM in Ireland	41
4.4 Enhancing Knowledge Bases at the Intersection of CCA & DRM in Ireland	42
5. Insights from the literature	43
5.1 The Next Steps	44
References	45

# Acknowledgements

The authors acknowledge review and comments made to earlier versions of this document by Tommaso Natoli (Irish Research Council/Marie Skłodowska-Curie Action CAROLINE Fellow in the School of Law of the University College Cork) and the EPA appointed Steering Group to the project.

# **Table of Acronyms**

AGS	An Garda Síochána	
AR	Assessment Report	
ССА	Climate Change Adaptation	
СОР	Conference of the Parties	
DAFM	Department of Agriculture, Food and the Marine	
DCCAE	Department of Communications, Climate Action and the Environment	
DEASP	Department of Employment Affairs and Social Protection	
DF	Department of Finance	
DHPLG	Department of Housing, Planning and Local Government	
DJE	Department of Justice and Equality	
DM	Disaster Management	
DRCD	Department of Rural and Community Development	
DRM	Disaster Risk Management	
DRR	Disaster Risk Reduction	
DTTS	Department of Transport, Tourism and Sport	
GHG	Greenhouse Gases	
GTF	Government Task Force	
HSE	Health Service Executive	
IFRC	International Federation of Red Cross and Red Crescent Societies	
IPCC	Intergovernmental Panel on Climate Change	
IRCG	Irish Coast Guard	
LGD	Lead Government Department	
MEM	Major Emergency Management	
NAF	National Adaptation Framework	
NDC	Nationally Determined Contributions	
OPW	Office of Public Works	
PA	Paris Agreement (under the UNFCCC)	

PRAs	Principal Response Agencies		
SDG	Sustainable Development Goals		
SEM	Strategic Emergency Management		
SFDRR	Sendai Framework on Disaster Risk Reduction		
TII	Transport Infrastructure Ireland		
UNDRR (UNISDR)	United Nations Office for Disaster Risk Reduction (formerly United Nations International Strategy for Disaster Risk Reduction)		
UNFCCC	United Nations Framework Convention on Climate Change		

# **Key Definitions**

The following definitions draw primarily from the definition presented in the relevant Irish frameworks. Where the definition of a term is not present in an Irish context, the international or EU definition has been used, and in some cases, both are used to exhibit how they correspond with one another.

Climate Change Adaptation	A change in natural or human systems in response to the impacts of climate change. These changes moderate harm or exploit beneficial opportunities and can be in response to actual or expected impacts (National Adaptation Framework, DCCAE, 2018; 98).
	The process of adjustment to actual or expected climate and its effects. In human systems, adaptation seeks to moderate or avoid harm or exploit beneficial opportunities. In some natural systems, human intervention may facilitate adjustment to expected climate and its effects (IPCC, 2014b).
Climate Change Mitigation	Describes action to reduce the likelihood of an event occurring or reduce the impact if it does occur. This can include reducing the causes of climate change (e.g. emissions of greenhouse gases) as well as reducing future risks associated with climate change (National Adaptation Framework, DCCAE, 2018; 99).
	A human intervention to reduce emissions or enhance the sinks of greenhouse gases (IPCC, 2018).
Disaster Risk Management	Application of disaster risk reduction policies and strategies to prevent new disaster risk, reduce existing disaster risk and manage residual risk, contributing to the strengthening of resilience and reduction of disaster losses (UNDRR Terminology, 2017).
Disaster Risk Reduction	Disaster risk reduction is aimed at preventing new and reducing existing disaster risk (exposure, hazard or vulnerability), and managing residual risk, all of which contributes to strengthening resilience and therefore to the achievement of sustainable development (IPCC, 2014a; UNDRR Terminology, 2017).
Exposure	The situation of people, infrastructure, housing, production capacities and other tangible human assets located in hazard-prone areas (UNDRR Terminology, 2017).
Emergency	An event which, usually with little or no warning, causes or threatens to cause death, serious injury, serious disruption to essential services, the economy or critical infrastructure, significant damage to property or the environment, and which requires the activation of national resources to ensure an effective coordinated response and recovery (SEM Framework, DoD, 2017; 2).
Hazard	Any phenomenon with the potential to cause direct harm to members of the community, the environment or the physical

	infrastructure, or being potentially damaging to the economic and social infrastructure (SEM Framework, DoD, 2017; 16).	
	A process, phenomenon or human activity that may cause loss of life, injury or other health impacts, property damage, social and economic disruption or environmental degradation (UNDRR Terminology, 2017).	
Impact	The consequences of a hazardous event actually happening, expressed in terms of a negative impact on human welfare, economic activity, environmental welfare or societal structures (SEM Framework, DoD, 2017; 16).	
Loss and Damage	The harms caused by anthropogenic climate change (UNFCCC, 2003).	
Resilience	Community resilience: The ability of a system, community or society exposed to hazards to resist, absorb, accommodate, adapt to, transform and recover from the effects of a hazard in a timely and efficient manner, including through the preservation and restoration of its essential basic structures and functions through risk management (SEM Framework, DoD, 2017; 18). The consequences of a hazardous event actually happening, expressed in terms of a negative impact on human welfare, economic activity, environmental welfare or societal structures (National Adaptation Framework, DCCAE, 2018; 100).	
Risk	The combination of the likelihood of a hazardous event and its potential impact (SEM Framework, DoD, 2017; 16).	
Vulnerability	The predisposition to be adversely impacted by the effects of climate change (IPCC, 2014a). The conditions determined by physical, social, economic and	
	environmental factors or processes, which increase the susceptibility of a community to the impact of hazards (UNDRR Terminology, 2017).	

#### **Executive Summary**

The scope of the present literature review is under the remit of a wider project entitled *Enhancing* Integration of Disaster Risk and Climate Change Adaptation into Irish Emergency Planning which is funded under the Environmental Protection Agency's Climate Topic 3 funding call. The project, which began in March 2020, is due to run until March 2021. The objective of the project is to help institutions responsible to further 'climate-proof' emergency planning and risk management systems in Ireland to the increasing risk of extreme hydrometeorological events, by addressing national policy and decision-making processes, as well as local and regional planning and response mechanisms. In order to do this, the project sets out three deliverables, the first of which is the present initial literature review. The second, is an extensive mapping of key stakeholders in both CCA and DRM institutions, assessing how SEM structures and functions may be affected based on a range of scenarios and timeframes. The third deliverable is a guidance roadmap, which will – with the aid of three case studies – examine the understanding of key stakeholders of how climate-related factors are currently affecting the vulnerability and resilience capacity of communities and sectors to short- to mediumterm extreme events and how these are likely to be affected by climate change. Reflecting the impacts of physical distancing arising from the COVID-19 restrictions, the project will identify appropriate methods for engaging with communities and key local, regional and national stakeholders. Reports from these stakeholder engagements will feed into the project roadmap by elaborating community understanding of current and future climate risks and impacts and the implications for emergency planning in the context of existing policy instruments and approaches, aiming to build a more climate and disaster resilient Ireland.

#### What's the issue?

The Government's National Risk Assessment 2019 highlighted climate change as one of the nation's **highest strategic priorities**. The public rightly expects effective action to manage this risk. Significant research and policy development work has occurred in Ireland to identify the interlinkages between Disaster Risk Reduction (DRR) and Climate Change Adaptation (CCA). The interlinkages and dependencies are well recognised and are reflected in a range of Irish policies such as the Strategic Emergency Management (SEM) Framework, National Adaptation Framework and Local Authority and Sectoral Adaptation Strategies. However, there is much to be gained from improving the alignment and integration of DRR and CCA, reducing risk and increasing resilience. Venton & La Trobe (2008) have observed that DRM measures will lessen climate impacts, and adaptation measures will reduce disaster risk. Moreover, the similar aims and agenda of DRM and CCA would benefit both areas, increasing awareness and knowledge development in both paradigms.

The present literature review explores how the various policy strategies and approaches that address aspects of DRR and CCA interact with each other, and the various gaps and challenges that exist in

v

integrating both areas. Moreover, it highlights various good practices and opportunities going forward to advance integration between both sectors. Although there has been recognition of the mutual benefits in integrating both areas, which is apparent throughout Irish policy and legislation, achieving this integration remains a gap. The key issues that have emerged from this initial review of the literature are outlined below.

#### **Issues that merit further investigation:**

- Terminology and fundamental concepts often differ across sectors, leading to miscommunication and discordance in actions and policies. This is particularly relevant in the term 'resilience'. In a disaster risk management context, the concept of resilience is normally used in the context of returning to normal, or pre-disaster conditions. However, for climate change adaptation, resilience relies on advancing forward, with returning to normal regarded as regressive.
- There is room for a stronger identification of the most vulnerable groups and communities, with their specific needs and interests prioritised, and resilience to future shocks supported. Moreover, their presence and participation in preparation and response processes is integral.
- 3. The importance of stronger coordination and information-sharing mechanisms for the horizontal integration of DRM and CCA at a national and sub-national level.
- 4. A stronger focus on the Sendai Framework's *'Build Back Better'* during disaster response and recovery, to add additional resilience of infrastructure and communities, particularly those facing a heightened threat of climate-induced severe events.
- Taking into account the financial repercussions of unmitigated climate risk, consideration should be given to the alignment of financial and economic systems to support CCA and DRM planning and implementation leading to more efficient use of resources (EEA, 2017).
- 6. Acknowledging the impact of climate change and disasters in all sectors, how best can CCA and DRM methods be mainstreamed to promote a multi-sectoral approach to both areas.

This review is the first step in the project deliverables, providing an overview of the relevant literature, as well as highlighting examples of good practices and potential barriers in enhancing the integration of disaster risk and climate change adaptation into Irish emergency planning. It provides background on material at an international, European and national level, and how Ireland interacts with key legal and policy instruments. Moreover, it highlights the need for enhanced integration, setting the foundation for the completion of future project deliverables. It should also be noted that as this publication was being finalised, a new coalition Government was sworn in, and the issue of climate change was identified as a clear priority in the Programme for Government. Any changes to the existing policies and structures will be therefore be closely tracked over the course of the current project.

## **Introduction & Background**

Global trends, including population growth, poverty reduction, unsustainable urbanisation and poor land management, ecosystem degradation and climate change have led to an increase in the frequency, intensity and impact of disaster risks over the last number of decades (EC, 2016). A study conducted in 2017, and published in the Lancet, observed that weather-related disasters could potentially affect two-thirds of the European population by the year 2100, causing fifty times the number of fatalities compared to today (Forzieri et al, 2017; 1). The Irish Government acknowledges that the projections of rising sea levels, extreme weather, pressure on water resources, flooding, displacement and heightened risk of diseases will undoubtedly have an impact on every individual, household and community across Ireland (Government of Ireland, 2019). Moreover, there is growing recognition of the effect climate change will have on the current emergency management structures. Highlighting these issues and recognising the mutual benefits of working collaboratively, will strengthen disaster risk management (DRM), while concurrently allowing for a more comprehensive approach to climate change adaptation (CCA).

As understanding of the relationship between CCA and DRM grows, there has been increasing recognition of the common objectives that both fields share – not only the importance of adaptation and risk reduction in climate action, but the co-benefits when integrated with one another. This is supported by the 2014 IPCC Report, which indicates that disaster risk management and adaptation policies – if well-coordinated – can be reinforcing and supportive of one another, leading to positive outcomes for both spheres. Failing to consider the predicted increasing intensity and frequency of natural hazards as a result of climate change has the potential to lead to maladaptation (Banwell et al, 2018; 9).

Ireland, as a nation, has increased its resilience to climate change through its legislative frameworks and planning requirements to adapt to climate impacts and manage disaster risk. As stipulated in the National Adaptation Framework (2018):

"There is a growing recognition at EU/international level of the need for greater integration of emergency planning (particularly disaster risk reduction) and climate change adaptation. As also indicated earlier, this has already begun in Ireland. Under this Framework, it is foreseen that these relationships will continue to strengthen over time" (National Adaptation Framework, 2018; 69). Although there has been recognition of the mutual benefits in integrating both DRM and CCA, which is apparent throughout Irish policy and legislation, achieving this integration remains a challenge. The draft fiche for Ireland in the European Commission's Directorate General for Climate Action Preparedness Scoreboard finds that *"There is not an integration of [disaster risk reduction] and [Climate Change Adaptation] policies in Ireland, although there are plans to promote it"* (Shine, 2018; 14). This review aims to contribute to that promotion by highlighting these gaps and challenges, and outlining the potential opportunities for integration, moving forward toward a more climate and disaster resilient Ireland.

The present literature review is completed in unusual circumstances, with the emergency management system in Ireland facing its largest public health emergency to date. The COVID-19 crisis will undoubtedly have an impact on sustainable development and disaster risk management in Ireland. However, the current global situation illustrates the benefit of integrated frameworks at a national and sub-national level, and vertical alignment from the international to the local. The threat of climate change has not disappeared with the COVID-19 pandemic, but rather exhibits their resemblance with one another. The pandemic has shown how vulnerable we are to a global emergency, and that unless sufficient planning and preparedness in anticipation of these events are not taken, we will once again face global consequences that – according to the 2018 IPCC Report – will be unsurmountable. The most recent National Risk Assessment Overview (2019) affirms this and recognises that "planning and preparedness are critical to help mitigate the impact of a pandemic. When the next pandemic occurs, it will require a whole-of-government response to ensure that threat to public health and disruption of services and society are minimised" (Government of Ireland, 2019; 64).

Similar to the reasoning for integrating CCA and DRM to pursue the goal of community resilience, the COVID-19 recovery period addresses a similar aim, with former UN Deputy Secretary-General Lord Mark Malloch Brown, forecasting that the term *'resilience'* will be the buzzword for the COVID-19 recovery. Similarly, Raul Salazar, head of UNDRR's Regional Office for the Americas and the Caribbean, speaking on the pandemic, expressed how the current global situation has conveyed the importance of national governments integrating different sectors, ensuring they work together rather than in silos. He acknowledged the importance of a systemic approach to prevention activities to ensure that disaster risk is reduced and lives saved (UNDRR, 2020).

Section one of this review provides an overview of climate change adaptation and disaster risk management at a national, European and international level outlining various frameworks and instruments such as the EU Civil Protection Mechanism (2013), the UN Sendai Framework for Disaster Risk Reduction (2015-2030) and the Paris Agreement (2015), as well as the historical

alignment and disparities among both sectors. The first section also describes the concepts of resilience and vulnerability as they stand in international policies and frameworks.

Section two explores disaster risk management and reduction on a local and national level, outlining the various frameworks and policy instruments, such as the Irish Strategic Emergency Management (SEM) Framework (2017). Following on from this, it discusses climate change adaptation in greater detail, along with the various institutional frameworks and policies applicable in Ireland, such as the Climate Action Plan (2019) and the National Adaptation Framework (2018). Moreover, it addresses the concept of resilience that is present within CCA & DRM in Ireland, as well as the institutional and policy linkages and potential barriers to coordination and integration of disaster risk and climate change adaptation into emergency planning. Additionally, section two identifies the key stakeholders and actors in Ireland. An integral component of the present research relates to a mapping of stakeholders, and stakeholder engagements, to enrich the lessons learned from the research and inform the outputs of the project. Insights from various key actors and stakeholders on the case studies which will be undertaken during the project will ensure that lessons learned can be applied in practice.

Section three then explores the various challenges, gaps and opportunities of enhanced integration of CCA and DRM in Ireland. It does so by highlighting notable examples of integration, followed by identifying certain gaps and potential barriers. In an attempt to bridge those gaps, section four highlights various good practices and examples of integration in an international, EU and local context. This will better inform national/sub-national policy and identify key themes that could potentially strengthen the collaboration between the existing Climate Change Adaptation (CCA) and Disaster Risk Management (DRM) frameworks for Ireland. Section five provides an overall conclusion for the literature review, while outlining next steps for the project.

# 1. CCA & DRM – Making the Connections

The call for greater integration of Climate Change Adaptation (CCA) and Disaster Risk Management (DRM) frameworks has increased in recent years and is now considered a global priority, according to the UNDRR Global Assessment Report on Disaster Risk Reduction (2019). Efforts to reduce disaster risk while simultaneously adapting to climate change is of great importance in both an international and EU context (Government of Ireland, 2019). As highlighted by the European Environmental Agency:

"The impacts of weather- and climate-related hazards on the economy, human health and ecosystems are amplified by socio-economic changes and environmental changes. Efforts to reduce disaster risk and at the same time adapt to a changing climate have become a global and European priority." EEA Report (No 15/2017)

Both approaches seek to reduce exposure, vulnerability and risk of people and assets to a range of hazards. Integrating the approaches in legal and policy frameworks, coordination and implementation structures, financing arrangements, capacity development efforts and accountability mechanisms, among other settings, will have greatest impact on the reduction of loss and damage from disasters and climate change while increasing efficiency in use of resources.

#### 1.1 Historical alignment and disparities between CCA and DRM

The historical development of CCA and DRM has had an impact on how both areas communicate and collaborate with one another. The two approaches have partially diverging backgrounds, methodologies and scopes of action (Natoli, 2019; 11). With the development and publication of the IPCC Special Report on Managing the Risk of Extreme Events and Disasters to Advance Climate Change Adaptation (SREX Report, 2012), as well as with the activation of the special programme on loss and damage of the UN Framework for the Convention on Climate Change (UNFCCC, 2012), greater emphasis has been placed on fostering coherence and alignment between DRM and CCA at an international level. Although in recent years there has been an attempt to bring both frameworks closer together, fragmentation persists, with some notable examples of a lack of coherence.

#### 1.2 Disaster Risk Management (DRM) and Disaster Risk Reduction (DRR)

The two dominant trends that are present in contemporary disaster risk management (DRM) are increasing the resilience of individuals, families and communities in regard to their exposure and vulnerability to hazards, and concurrently strengthening the local, regional and national disaster

prevention, preparation and response systems. The first has emphasised the importance of disaster risk reduction, with the concept of risk as central to disaster management. The latter, seeking to strengthen DRM systems, focuses on establishing various normative frameworks to underpin disaster management, adopting a risk management focus rather than disaster response approach (McDermott & Gibbons, 2015).

Disaster risk reduction (DRR) has become an established component of DRM, as efforts to increase long-term resilience while responding to short-term risk increases, thus incorporating climate change and natural hazards. For the purpose of this review, the term DRM will be employed, acknowledging the importance of DRR as a global agreed policy to effective disaster risk management (UNDRR, 2020). At a global and European level, a high priority is now placed on implementing an integrated risk approach by taking into account the full disaster management continuum (prevention/mitigation, preparedness, response and recovery) which further considers climate change as a driver of risk (EEA Report, no.15/2017; 16). This is reflected at the national level, with the Irish Strategic Emergency Management Framework adopting a five-phase systems approach (hazard, mitigation, planning and preparedness, response, and recovery) based on a continuous cycle of activity (DoD, 2017).

Disaster risk refers to the potential loss of life or injury, or the destruction or damage caused to assets in a system, society or community during a specific period of time, determined probabilistically as a function of hazard, exposure, and vulnerability (EEA no.15/2017). Risk is determined as the consequence of the interplay between a hazard and certain factors that may leave people and places exposed and vulnerable (UNISDR, 2015).





# 2. CCA & DRM in Ireland – An Overview

## 2.1 An Overview of Relevant Legal & Policy Frameworks in Ireland

Ireland's national laws, policies and practices for disaster management and climate change adaptation draw from and are aligned with three key international frameworks, as set out below.

	Risk Management	Climate Change	Sustainable Development
International Sendai Framework on DRR (2015-2030)		UNFCCC / Paris Agreement	UN Agenda 2030 & the SDGs
National	National Risk Assessment 2019: Overview of Strategic Risks (2019) and National Risk Assessment (2017)	Climate Action Plan 2019 To Tackle Climate Breakdown (2019)	A Better World: Ireland's Policy for International Development (2019)
	Strategic Emergency Management Framework (2017)	National Adaptation Framework: Planning for a Climate Resilient Ireland (2018)	Ireland: Voluntary National Review (2018)
	Major Emergency Management Framework (2006)	Local Authority Climate Change Adaptation Guidelines (2018)	Our Sustainable Future: A Framework for Sustainable Development in Ireland – Progress Report (2015)
		National Mitigation Plan (2017)	Our Sustainable Future: A Framework for Sustainable Development in Ireland (2012)

Table 1: Overview of key international and national legal and policy frameworks

The need for coherence between the different legal and policy frameworks at the international level, in particular covering the three topics of disaster risk, climate change and sustainable development, is well recognised (Sandholz, 2020), and can be augmented by the vertical alignment between the international, regional, national and local levels (Cubie & Natoli, forthcoming). This allows for the sharing of knowledge and expertise regarding legal principles and operational experience, and promotes a coordinated approach to risk management, as visualised in Figure 3 below.



Figure 2: Coherence and alignment of key frameworks.

#### 2.2 Disaster Risk Management

#### National Context

In its approach to disaster risk management, Ireland incorporates a 'whole of government approach', encompassing emerging issues or trends which may be best addressed elsewhere, and considering expert opinion and *"formulating and agreeing national practices"* (DoD, 2017; 26). Legal provision for disaster management in Ireland can be found across a wide range of legislative domains. Various policy documents, such as the Strategic Emergency Management Framework and the Framework for Major Emergency Management, provide a shared understanding of roles and responsibilities from the national level, to a local and regional level (McDermott & Gibbons, 2015; 367).



Figure 3: Five stage systems approach (DoD, 2017)

The three principles which underpin all levels of emergency planning are: i) an 'all-hazards' approach; ii) subsidiarity, with initial emergency response at the nearest, local level; and iii) effective coordination at the local, national and regional level (DoD, 2017; 3). Responsibility for disaster management on a national level lies with Lead Government Departments (LGD's), which are identified in the SEM Framework for 50 different types of emergency situation. The LGD role includes risk assessment, planning and preparedness, prevention, mitigation, response, and recovery.

#### Response and Risk Management Authorities in Ireland

The National Directorate for Fire and Emergency Management (NDFEM) was created within the Department of Housing, Planning and Local Government in 2009 and it has the designated agency responsible for leading and supporting development of the full range of emergency management functions for which the Department is responsible. For example, the NDFEM is tasked with receiving and assessing weather warnings and reports on weather events around the country (DHPLG, 2019).

#### Local Level Coordination

On a local level, the responsibilities for identified emergencies are set out in the Major Emergency Management Framework (2006). The Framework specifies the arrangements to organise responses to emergencies in a coordinated manner, including on a local and regional level, and designates local authorities as the lead agencies in government response. It includes mechanisms for linking the Principal Response Agencies (PRAs) and other stakeholders in the management of emergencies (Soldi, 2016).

The three Principal Response Agencies (An Garda Síochána, Health Service Executive and the Local Authorities) are responsible for ensuring procedures for preparing for and responding to an emergency are detailed in the relevant emergency plans. The Framework declares itself as an overall response measure *"as soon as a major emergency is declared"*, thus affirming its position as predominantly a response mechanism to emergencies (DHPLG, 2006; 14). In line with the Framework, PRA's are



Figure 4: Regional Major Emergency Management map (MEM, 2006)

mandated to adopt Major Emergency Plans, updated every year, to plan for and respond to events. The country is divided into eight regions for civil protection purposes; referred to as 'Major Emergency Management Regions'. For each region, there is an inter-agency Regional Steering Group (RSG) and a Regional Working Group (RWG), including managerial and operational staff of the PRA's. Once an emergency is declared, each PRA is required to inform its parental central government department of the emergency (McDermott & Gibbons, 2015). County Councils and City Councils are among the Principal Response Agencies appointed to manage emergencies (Soldi, 2016). The SEM Framework identifies Lead Government Departments (LGD) with the mandate for risk assessment, planning and preparedness, prevention, mitigation, in addition to response and recovery roles. In addition, other government departments and agencies are designated to provide either principal or other support to the LGD across each of the 50 major hazards identified in the SEM Framework. DHPLG is the LGD for the major climate-induced hazards facing Ireland, specifically severe weather, flooding and fires, with each hazard requiring contributions from different supporting departments and agencies.

#### European Context

The EU Civil Protection Mechanism (2013) addresses significant issues relating to disaster risk management. It explicitly tackles these issues through mapping key risks, undertaking prevention and preparedness missions, supporting research and knowledge sharing, peer reviews and enhancing international cooperation (EU Civil Protection Mechanism, 2013). The mechanism requires countries to conduct comprehensive multi-hazard risk assessments (EEA Report, no.15/2017). Ireland, along with 32 other countries, encompassing both EU Member States and other participating states, engage within the Civil Protection Mechanism. Ireland conducts a national risk assessment every five years in compliance with the mechanism (DoD, 2017). The EU approach to DRM is formed by a number of thematic instruments, central to which, is the EU Civil Protection Mechanism (EU, 2017; 29). Additionally, the EU Action Plan on the Sendai Framework for DRR 2015-2030 recognises the Sendai Framework as an opportunity to bolster EU resilience to shocks and disruption in the context of EU development. It highlights that translating the Sendai Framework into tangible actions should be done in accordance with other international agreements, including the 2030 Agenda for Sustainable Development and the 2015 Paris Agreement (EC, 2015).

#### International Context

There are a number of key international frameworks which have informed the development and implementation of national and regional DRM approaches. For example, the EU Civil Protection Mechanism draws heavily on principles set out at an international level, in particular the Hyogo Framework for Action 2005-2015 and its successor, the Sendai Framework for Disaster Risk Reduction (SFDRR) 2015-2030. The Sendai Framework was adopted at the third United Nations World Conference on Disaster Risk Reduction, with a primary aim to "prevent new and reduce existing disaster risk through the implementation of integrated and inclusive economic, structural, legal, social, health, cultural, educational, environmental, technological, political and institutional measures that prevent and reduce hazard exposure and vulnerability to disaster, increase preparedness for response and recovery, and thus strengthen resilience" (SFDRR, 2015; 12).

Although the Sendai Framework is not legally binding, several countries have highlighted the need for strengthening DRM worldwide (UNDRR, 2015), and Ireland was an active participant in the final negotiations of the Framework. At the Third World Conference on Disaster Risk Reduction in March 2015 – where the Sendai Framework was adopted – Ireland stated its support for a comprehensive global plan on disaster risk reduction and management that *"prioritises the most vulnerable, delivers results at country, regional and global levels, is complementary to other global processes with similar aims, is appropriately resourced, is accountable and will, overall, have the capability to guide the global response to disaster risk for the coming years*" (Government of Ireland, 2015). Although the SFDRR is not referenced by the Strategic Emergency Management Framework, the Irish structures do encompass various elements of the four priorities for action which form the central feature of the international framework.

### Ireland's Integration of SFDRR in the SEM Framework

The table below provides a summary of the inclusion of the Sendai Priorities for Action into the Irish SEM Framework, many of which will be discussed in more detail in the rest of this literature review.

Sendai Framework Priorities for Action		Strategic Emergency Management Framework	
1.	Understanding disaster risk	<ul> <li>Clear definition of risk and hazard, with a process for national risk assessment.</li> <li>Identifies and communicates emergency scenarios that could occur at a national and local level – as a result of natural, human-induced and climate hazards.</li> <li>Risk exposure and vulnerability of persons not specifically applied.</li> </ul>	
2.	Strengthening disaster risk governance to manage disaster risk	<ul> <li>Clear outline of roles and responsibilities to enable effective management of risk.</li> <li>Clear outline of arrangements and processes in place to enable delivery of national and regional level emergency management (DoD, 2017; 6).</li> </ul>	
3.	Investing in disaster risk reduction for resilience	<ul> <li>Resilience and community resilience a key component in SEM Framework.</li> <li>Investing and funding predominantly in relation to response and recovery.</li> <li>Expenditure requirements include immediate expense, long-term large-scale infrastructure, longer term assistance, longer term additional resilience measures (DoD, 2017; 29).</li> </ul>	

Table 2: Summary of Sendai Priorities for Action in Irish SEM Framework

4.	Enhancing disaster preparedness	• Proposals for funding projects to rehabilitate,		
	for effective responses and to		restore, or provide 'additional resilience' will be	
	'Build Back Better' in recovery,		"considered by the relevant funding	
	rehabilitation and reconstruction		department" (DoD, 2017; 29).	

To promote accountability of governments and other relevant stakeholders, the SFDRR sets out seven specific targets (SFDRR, 2015). Some of these guiding targets are contained within the Irish SEM Framework, although explicit reference is not made to the SFDRR. In 2018, the OECD released a country profile on Ireland's policy integration. It reported on a seminar held by the Government Task Force, Department of Foreign Affairs and Trade and UNISDR on introducing the SFDRR as Ireland determined its institutional roles and responsibilities as a signatory to the Framework and determined how effectively the SFDRR corresponded with the Irish National Adaptation Framework (OECD, 2019). In its Policy for International Development (2019), the Government affirmed that it intended to *"strengthen [our] approach to disaster risk reduction in line with the 2015 Sendai Framework and continue to track our spending"* (Government of Ireland, 2019; 17). However, there is no express institutional or policy linkage between Ireland's international development policy and its domestic disaster management policy.

#### 2.3 Climate Change Adaptation

Climate change is contributing to an increase in disaster risk, with the impacts – rising sea-levels and extreme weather – being observed first-hand in Ireland (Government of Ireland, 2019). Temperatures in Ireland have increased by about 0.8°C since 1900 (Nolan, 2015). Moreover, the mean annual sea surface temperature – as measured in Malin Head, Co. Donegal – is now more than 1°C higher than the long-term average (National Adaptation Framework, 2018; 26), with changes expected to continue and increase over the coming years. A report by Met Éireann published in 2013 succinctly outlined future projections for Irish weather in the coming years using global climate model simulations. Among the key estimations was an increase of 1.5°C by mid-century, an increase in storm wave heights, and up to 20% increase in frequency of heavy precipitation events – which will have an effect on river catchment hydrology (Gleeson, 2013; 7).

The impacts of climate change on sectors and regions across Ireland are dependent not only on climate-related hazards, but on current and future exposure and vulnerability of a diverse range of human, natural and socio-economic systems. It is therefore recognised that it is necessary to reduce and manage climate change risks through a combination of mitigation and adaptation focused responses (National Adaptation Framework, 2018; 32). In 2018, Ireland was ranked as the worst EU Member State based on its climate action (Burck et al, 2019), and had the highest national proportion of agriculture emissions (EPA, 2019). Although this is largely due to its role in supplying meat and

dairy products across Europe, Ireland also has a higher carbon footprint per head of the population in the electricity, transport and waste management sectors (Government of Ireland, 2019; 9).

#### National Context

The country's first National Mitigation Plan was published in 2017 and works in tandem with the National Adaptation Framework for managing greenhouse gas emissions and achieving the 2050 target of a decarbonised economy (DCCAE, 2017; 9). The National Adaptation Framework, published in 2018, sets out current progress and provides guidelines for adaptation on a national level. Its subtitle: 'Building resilience to climate change' illustrates the role of adaptation in reducing vulnerability and enhancing resilience. It complements the required mitigation actions by addressing the resilience challenge and specifies strategies for adaptive measures in various sectors and local authorities. Furthermore, it intends to "reduce the vulnerability of the state to the negative effects of climate change and to avail of any positive effects that might occur" (DCCAE, 2018; 9). At the local level, as part of the country's National Adaptation Framework, all local authorities are required to produce a Climate Adaptation Strategy in line with the framework. Additionally, the Framework identifies 12 sectors across seven government departments and agencies for the development of climate adaptation strategies – including actions to be implemented at a local government level (Cork County Council, 2019). Support for the drafting of these adaptation plans exists via the Sectoral Planning Guidelines for Climate Change Adaptation (2018) and the Local Authority Adaptation Sectoral Development Guidelines (2018).

Sector	Parent Department	
Seafood	Department of Agriculture, Food & the Marine	
Agriculture	Department of Agriculture, Food & the Marine	
Forestry	Department of Agriculture, Food & the Marine	
Biodiversity Department of Culture, Heritage & the Gaeltacht		
Built & Archaeological Heritage Department of Culture, Heritage & the Gaeltacht		
Transport Infrastructure         Department of Transport, Tourism & Sport		
Electricity and Gas Networks	Department of Communications, Climate Action & Environment	
Communications Networks	Department of Communications, Climate Action & Environment	
Flood Risk Management	Office of Public Works	
Water Quality   Department of Housing, Planning & Local Govern		
Water Services Infrastructure Department of Housing, Planning & Local Governmen		
Health Department of Health		

Table 3: Sectoral responsibilities (Cork County Council, 2019)

The Irish government has undertaken a progressive step on climate action in Ireland and plans to move toward decarbonisation with the Climate Action Plan to Tackle Climate Breakdown (2019). The Plan emerged after the Irish Government became the second country – following the United Kingdom – to declare a climate emergency (House of the Oireachtas, 2019). The Plan comprises a comprehensive proposal to reduce greenhouse gas emissions and foster climate resilience and provides a roadmap for all stakeholders to work together in creating a *"resilient, vibrant and sustainable country"* (Government of Ireland, 2019; 8). The Plan acknowledges that the potential impacts and costs of inaction to the effects of climate change are significant, and thus requires robust adaptation measures in all areas of governance. Moreover, it recognises both the Paris Agreement and the UN's 2030 Agenda for Sustainable Development as integral in climate action. The Plan also commits to bring forward a new Climate Action (Amendment) Bill to amend the Climate Action and Low Carbon Development Act (2015). The new Bill aims to act as a strengthened statutory framework for long-term planning in placing Ireland on a trajectory to be net zero in emissions by 2050 (DCCAE, 2019).

Ireland's targets are binding at an EU level, under the EU Efforts Sharing Decision (ESD). The ESD forms part of the EU's climate and energy policy framework for 2020, where national emission targets for 2020 are set, expressed as percentage changes from 2005 levels (EC, 2020). As per its 2020 targets, Ireland was legally bound to deliver 16% of its final energy requirements from renewable sources, 10% renewable energy in transport, and a 20% decrease in its emissions from its 2005 figure, stated in the National Mitigation Plan (DCCAE, 2017; 36). Although it did not meet this target, the Climate Action Plan (2019) with its decarbonisation pathway to 2030, is in line with the aim of envisioning a net zero target in Ireland by 2050 (Government of Ireland, 2019).

#### Local Level Coordination

As mentioned above, the National Adaptation Framework calls for the implementation of local adaptation strategies. For many authorities, this strategy was the first of its kind so the Local Authorities Adaptation Guidelines were developed to help counties and regions draft their own strategy. The guidelines highlight the necessity for adaptation strategies to be mainstreamed into local authority plans and policies to 'climate-proof' existing plans and policies (DCCAE, 2018). A network of four Climate Action Regional Offices (CARO's) have been set up to support this requirement, facilitating collaboration on a regional partnership approach to adaptation strategies, to coordinate and advise local authorities on adaptation strategies and contribute to national dialogue on climate action on a regional and local level. Additionally, the guidelines suggest that CARO's and local authorities *"explore synergies and collaboration that may be mutually beneficial"* (DCCAE, 2018; 20). The grouping of regions is as follows:

Climate Action Regional Office	Local Authorities in CARO Region	CARO Lead Authority
Atlantic Seaboard North	ntic Seaboard North Donegal, Sligo, Mayo, Galway City & County	
Atlantic Seaboard SouthClare, Limerick, Kerry, Cork City & County		Cork County Council
Eastern & Midlands	Carlow, Cavan, Kildare, Kilkenny, Laois, Leitrim, Longford, Louth, Meath, Monaghan, Offaly, Roscommon, Tipperary, Waterford, Westmeath, Wexford, Wicklow	Kildare County Council
Dublin Metropolitan	South Dublin, Fingal, Dun- Laoghaire-Rathdown, Dublin City	Dublin City Council

Table 4: Overview of Climate Action Regional Offices

Local authorities are at the intersection of emergency response and climate adaptation efforts, with the ability to prepare for and respond quickly and efficiently to disasters and potential risks, combined with local knowledge. From building flood defences to coordination of first responses, city and county councils are on the frontline of the climate crisis (LGMA, 2020). These experiences and knowledge mean that local government are in a position to act as a "catalyst for much wider change" in responding to short-term challenges while building long-term resilience (Clarke & O'Donoghue-Hynes, 2020; 8). The Climate Action Plan (2019) additionally acknowledges the instrumental role of local authorities, stating that: "local authorities occupy a pivotal role in their respective communities and can act to demonstrate public sector leadership on climate action in their areas as well as key mobilisers of action at a local and community level" (Government of Ireland, 2019; 127). In this respect, local authorities are in a unique position to engage with relevant stakeholders. An example of this is the detailed Mayo County Council Adaptation Strategy, where a monitoring and implementation mechanism to keep track of progress includes a structured and substantive programme for the engagement of local authorities, local community, relevant non-governmental organisations and state sector bodies, particularly those who will be expected to play a role in implementation of actions set out in the strategy (Mayo County Council, 2019; 106-110). The role of local authorities in emergency response is equally as critical. For example, with extreme weather events expected to continue, local authorities have begun to invest more in flood defences in recent years to help offset the future impact of these events (LGMA, 2020).

#### European Context

In 2013, the European Commission adopted an EU Strategy on Adaptation to Climate Change. The strategy has three main objectives:

- Promoting action by Member States: Encouraging all Member States to adopt comprehensive adaptation strategies, with the EU providing guidance and funding for countries achieving this aim. The creation of the Irish National Adaptation Framework (2018) is in accordance with this objective set out by the European Commission.
- 2. Promoting better informed decision-making: By addressing gaps in knowledge about adaptation, and further developing the climate-ADAPT platform, the 'one-stop shop' for information on climate adaptation in Europe. The <u>Climate Ireland</u> website plays a similar role in providing climate adaptation information, data, projections and support to local authorities, organisations, businesses and the general public (Climate Ireland, 2020).
- 3. Promoting adaptation in key vulnerable sectors: Through fisheries, agriculture and cohesion policy ensuring infrastructure is made more resilient against disasters. Under the National Adaptation Framework, seafood, agriculture, biodiversity and forestry along with other vulnerable sectors are mandated to develop their own climate adaptation strategies.

#### International Context

Adopted in 2015 under the UN Framework Convention on Climate Change, the Paris Agreement on Climate Change is a binding international legal agreement which seeks to hold the increase in global average temperature to well below 2°C above the pre-industrial level, while recognising that keeping the temperature increase to below 1.5°C would significantly reduce the risks and impacts of climate change. The overall objective of the Agreement is to strengthen the global response to the threat of climate change, in the context of sustainable development and efforts to eradicate poverty, including by:

"Increasing the ability to adapt to the adverse impacts of climate change and foster climate resilience and low greenhouse gas emissions development, in a manner that does not threaten food production" (Article 2(1)(b)).

As a signatory to the Paris Agreement, Ireland has stated that it is committed in its efforts toward meeting the Paris Agreement Objectives (DCCAE, 2019). National progress on adaptation is reviewed and reported at the international level via meetings of the Conference of the Parties, submission of adaptation commitments under the Paris Agreement, and a review of national climate plans every five years.

At a European level, reporting to the Commission every four years is required on national adaptation plans and strategies to facilitate adaptation on climate change and the EU Adaptation Preparedness Scoreboard (DCCAE, 2018). The EU Preparedness Scoreboard includes an assessment of each Member State's adaptation strategies, as well as a horizontal assessment of the 28 country reports (European Commission, 2018). Meanwhile, at the national level, oversight of progress is provided by the completion of a review on the National Adaptation Framework every five years by DCCAE, and the submission of annual and periodic review reports by the Climate Change Advisory Council which was established under s.8 of the *Climate Action and Low Carbon Development Act 2015* (DCCAE, 2018).

To promote the sharing of good practices, the EU financed two relevant projects through the Horizon 2020 research framework:

- The **ESPREssO Project** (Enhancing Synergies for disaster PRevention in the EurOpean Union) which ran between 2015-2019 aimed to promote new ideas on what should act as a future roadmap for hazard research and policymaking (Zuccarro et al, 2018b; 11); and
- The <u>PLACARD Interchange</u> (PLAtform for Climate Adaptation and Risk reDuction), which concluded in 2020, which aimed to facilitate a space for connection between those who are working in climate change adaptation and disaster risk reduction areas (PLACARD Interchange; 2020).

Publications arising from these projects provide insightful knowledge and practical guidance for Ireland on integrating national structures and policies addressing CCA and DRM, and reflect the need to "*improve the exchange of good practice*" between those responsible for the design of monitoring, reporting and evaluating of CCA and DRM measures (EEA, 2017; 141).

#### 2.4 Ireland and the UN 2030 Agenda for Sustainable Development

The UN 2030 Agenda for Sustainable Development places both CCA and DRM at the heart of sustainable development, as set out in the 17 Sustainable Development Goals (SDGs) (EEA, 2017a). The adoption of three key international agreements in 2015 (i.e. Sendai, Paris and the SDGs) underpinned the need to strengthen coherence at the international level between development, climate change and DRM which had often remained in distinct disciplinary silos. The clear interlinkages between the three agreements are set out in Figure 5 below.



Figure 5: Linkages between international frameworks (NAP Global Network Webinar 2020)

Crucially, all three agreements depend on each other's successful implementation – referencing each other and, according to the PLACARD Report, *"incentivising coherent implementation"* (Leitner et al, 2020; 29). It is therefore important to note that at the national level, the SDGs are a key focal point in both the National Adaptation Framework (2018) and the Climate Action Plan (2019).

Ireland, in its commitment to the SDG's, acknowledges Goal 13, which calls on countries to take urgent action to combat climate change and its impacts (UN, 2015). Ireland supports sustainable development at the domestic level through its 'Our Sustainable Future' framework, launched in 2012, with the most recent progress report published in 2015 (Government of Ireland, 2015). Disaster risk reduction touches on around 25 specific targets and are a key focus in 10 of the 17 Sustainable Development Goals – firmly highlighting the role of disaster risk management as a "core development strategy" (Leitner et al, 2020; 29). In particular, SDG Goal 11(b) aims to substantially increase the number of human settlements and cities adopting and implementing integrated policies, including "adaptation to climate change, resilience to disasters, and to develop and implement, in line with the Sendai Framework for Disaster Risk Reduction 2015-2030, holistic disaster management at all levels" (UN, 2015). The Strategic Emergency Management Framework, however, does not

reference sustainable development at any point, despite the key role that it plays in enhancing resilience.

#### 2.5 Mapping of Key Actors and Stakeholders in Ireland

"Decision-making that increases resilience to climate change impacts is not made in a vacuum, but rather, is set within a diverse social, economic and cultural landscape. It is, therefore, critical to engage the stakeholders that have an interest in both the decision-making process and outcomes in order that all needs are recognised" (Leitner et al, 2020; 46).

As outlined in the Introduction, stakeholder engagement is an integral component of fostering coordination and coherence within the CCA & DRM sphere (Amaratuga et al, 2017; 45). There are many benefits from participatory stakeholder engagements, such as sparking new ideas or gaining insight into perspectives on solutions that may have previously been overlooked. Engaging with stakeholders can bring forth local knowledge – which could be vital for identifying areas of vulnerability and possible strategies to manage risk effectively, as well as providing legitimacy to research (Leitner et al, 2020; 46). Due to the similar challenges and barriers CCA & DRM both face, an integrated and participatory approach may even be more effective. A multiple lens perspective is arguably more helpful than a single lens perspective, which may provide a narrow or one-dimensional solution (Leitner, 2020; 46). It has been emphasised by the ESPREssO Project that the major challenge is not identifying who key stakeholders are, but the allocation of tasks and responsibilities between these stakeholders, as roles and responsibilities may not be clearly defined (Amaratunga, 2017; 45). This, in turn, may lead to miscommunication among stakeholders when specific roles and liabilities become unclear. Therefore, unravelling the complexity of both areas is one of the first steps in enhancing integration between the two fields. To achieve this aim, Bojovic et al (2020) highlight that it is important to identify key actors within both areas, understand the interactions between them, analyse how they are connected to one another and what level of interaction they have, if any (ibid, 2020).

The National Adaptation Framework (2018) acknowledges the pivotal role that stakeholders play in contributing toward climate action and provides an extensive list of the key stakeholders contributing to climate adaptation in Ireland, including local authorities, civil society and the private sector. The Climate Change Advisory Council (CCAC) was established by Ministerial Order in 2016 under section 8 of the Climate Action and Low Carbon Development Act 2015, to aid in Ireland's transition to a low carbon, climate resilient and sustainable economy (Government of Ireland, 2015). Additionally, the Government made a commitment to establish a National Dialogue on Climate

18

Action (NDCA) to create community awareness, engagement and motivation to act, foster a flow of information, establish appropriate networks and providing for regular input (DCCAE, 2018). In addition to the NDCA, the Citizens' Assembly provides another avenue for stakeholder engagement (EC, 2015). However, in the European Commission's Adaptation Preparedness Scoreboard (2015), it was found that Ireland has made inadequate progress in terms of stakeholder involvement in the development of adaptation policy (DCCAE, 2015).

The SEM Framework identifies the key intergovernmental and interorganisational relationships, defining stakeholder relationship roles and responsibilities and outlining recovery phases and timelines. The Government Task Force (GTF) on Emergency Planning coordinates and oversees the emergency management policy and activities of all Government Departments and Agencies under their aegis and facilitates coordination of emergency management between Departments and Agencies (DoD, 2017). The GTF is chaired by the Minister for Defence and consists of senior representatives of all Departments, the Health Service Executive, An Garda Síochána, the Defence Forces, the Health and Safety Authority, the Revenue Commissioners, Met Éireann, the Environmental Protection Agency, the Office of the Government Chief Information Officer, Civil Defence, the Office of Public Works, the Irish Coast Guard and other Agencies as appropriate. The GTF Sub-Groups address emergency matters to minimise potential consequences of emergencies (DoD, 2017). The LGD is the mandated department responsible to coordinate all national level activity for its assigned emergency (DoD, 2017), along with supporting departments and agencies. By way of example, the following table outlines the stakeholders involved in four of the 50 emergency management scenarios which pose a high level of risk in Ireland:

Emergency type	Lead Government Department (LGD)	<b>Principal Support</b> and other support roles <sup>1</sup>	Remarks
Severe weather Storms/High Winds Thunder/Lighting Flooding/Snow/Ice Low/High Temperatures Drought Coastal Erosion	Department of Housing, Planning & Local Government (DHPLG)	<ul> <li>Local Authorities</li> <li>Dept of Housing, Planning &amp; Local Government</li> <li>An Garda Síochána (Dept of Justice &amp; Equality)</li> <li>HSE (Dept of Health)</li> <li>Met Éireann</li> <li>Civil Defence (Dept of Defence)</li> <li>Other Support Roles:</li> <li>Defence Forces (DoD), DCCAE, Environmental Protection Agency (DCCAE)</li> <li>Transport Infrastructure Ireland (Dept of Transport, Tourism &amp; Sport)</li> <li>Dept of Agriculture, Food &amp; the Marine</li> <li>Irish Coast Guard (DTTS)</li> <li>Office of Public Works (Dept of Public Expenditure &amp; Reform)</li> <li>Dept of Employment Affairs &amp; Social Protection</li> <li>Dept of Rural &amp; Community Development</li> </ul>	<ul> <li>Local Authority: Lead Agency as per Framework for Major Emergency Management.</li> <li>IRCG also has statutory powers re movement of shipping in and out of harbours &amp; anchorages during severe weather.</li> <li>DEASP has a Support Role under Humanitarian Assistance Scheme.</li> </ul>
Flooding	Department of Housing, Planning & Local	<ul> <li>Local Authorities</li> <li>HSE (Dept of Health)</li> <li>An Garda Síochána (Dept of Justice &amp; Equality)</li> </ul>	<b>Local Authority:</b> Lead Agency as per Framework for Major Emergency Management.

Table 5: Key stakeholders in selected emergency management scenarios (SEM, 2017)

<sup>&</sup>lt;sup>1</sup> A Principal Support Role is one that is explicitly mentioned in a Department's emergency plans. Other support roles include non-specific assistance, which may be requested from any Department or Agency in an emergency (SEM Framework, P. 6)

	Government (DHPLG)	<ul> <li>Office of Public Works (DPER)</li> <li>Met Éireann (DHPLG)</li> <li>Environmental Protection Agency (DCCAE)</li> <li>Dept of Agriculture, Food &amp; the Marine</li> <li>Civil Defence (Dept of Defence)</li> <li>Dept of Employment Affairs &amp; Social Protection</li> <li>Dept of Finance</li> <li>ESB</li> <li>Other Support Roles:</li> <li>Coillte (DAFM) Waterways Ireland Irish Water DF (DoD)</li> <li>Irish Coast Guard (DTTS)</li> </ul>	<b>OPW</b> is responsible per Government Decision (September 2004) in respect of flood risk identification, mitigation and awareness. (Ref Govt Decision S180/20/10/0996). Note: OPW agreed to transfer the functions and responsibilities in relation to coastal protection and coastal flooding on 1 January 2009 from DAFM. <b>DEASP</b> Support Role under Humanitarian Assistance Scheme.
Pandemic Influenza and Other Public Health Emergencies	Department of Health (DoH)	<ul> <li>HSE (All members of the Health Threats Coordination Committee)</li> <li>Other Support Roles:</li> <li>Local Authorities (DHPLG) CD (DoD)</li> <li>Revenue Commissioners FSAI19 (DAFM)</li> </ul>	Potentially all Government Departments/Agencies may be affected and involved, because of the potential widespread impact of a pandemic or other Public Health Emergencies.
Environmental Pollution	Department of Housing, Planning & Local Government (DHPLG) or Department of Transport, Tourism and Sport (DTTS) if at sea	<ul> <li>Local Authority (DHPLG</li> <li>Environmental Protection Agency (DCCAE)</li> <li>Dept of Agriculture, Food &amp; the Marine</li> <li>Health Service Executive (Dept of Health)</li> <li>Office of Public Works (Dept of Public Expenditure &amp; Reform)</li> <li>Irish Coast Guard (DTTS) (Marine incidents)</li> <li>Other support roles:</li> <li>OPW (DPER) DF (DoD)</li> </ul>	<b>Local Authority:</b> Lead Agency as per Framework for Major Emergency Management. Should the incident occur at sea, the lead may pass to DTTS and IRCG.

The involvement of both governmental and non-governmental stakeholders is deemed integral to emergency planning under the Sendai Framework. The Sendai Framework in Guiding Principle (d) states that DRR requires an all of society engagement, with *"inclusive, accessible and non-discriminatory participation, paying special attention to people disproportionately impacted by disasters"* (UNDRR, 2015; 8). In the UNDRR guide on implementing the Framework, it highlights that the designing and implementation of disaster preparedness plans cannot be done without women and girls, men and boys, people with disabilities, older persons, indigenous peoples, migrants or others with access and functional needs and vital capacities, their organisations and networks. Emergencies of any kind in Ireland are likely to have a cross-sectoral and cascading impact, with ties in agriculture, infrastructure, business, and flood risk management. By mapping stakeholders across all phases of DRM (prevention, preparation, response and recovery), it will be easier to ascertain ownership gaps.

The SEM Framework is clear in the identification of stakeholders in hazard-focused vertical silos. However, it does not reference important horizontally linked stakeholders in different, but related sectors, or non-governmental stakeholder organisations and groups. When decisions are made without understanding the effects in different sectors, there is a possibility of unintended consequences and counterproductive measures being implemented (Leitner et al, 2020). Therefore cross-sectoral communication is critical to foster solutions and enhance knowledge and understanding in various sectors to create more coordinated and coherent policy. Examples of good practice within Ireland include Donegal County Council which, in its adaptation plan, recognises the significance of sectoral adaptation plans to local adaptation, and vice versa. To ensure the necessary sectoral input is obtained, coordination between sectoral and local scale adaptation efforts are facilitated through CARO's (Donegal County Council, 2019; 15). Likewise, Cork County Council's Adaptation Plan aims to build and strengthen partnerships, by promoting inter-departmental communications and cooperation (Cork County Council, 2019; 46).

#### 2.6 Resilience and Vulnerability in CCA & DRM in an International Context

Both the climate change adaptation and disaster risk management sectors address prevention and reduction of risks of disasters by reducing vulnerability and increasing resilience. As visualised in Figure 6 below, resilience in the context of disasters is defined as:

"the ability of a system, community or society exposed to hazards to resist, absorb, accommodate, adapt to, transform and recover from the effects of a hazard in a timely and efficient manner, including through the preservation and restoration of essential basic structures and functions through risk management" (UNDRR Terminology, 2017).

*Figure 6: initial economic conditions impact on resilience and recovery (http://www.oedd.org/economic-resilience.htm)* 



The SFDRR necessitates a more people-centred preventative approach to disaster risk, with practices focusing on being *"multi-hazard and multisectoral, inclusive and accessible in order to be efficient and effective"* (SFDRR, 2015, 10). This is particularly important as vulnerability is the result of a range of economic, social, cultural, institutional and political factors that shape people's lives and their direct environment (Twigg, 2015). Those who are marginalised in society will face the steepest hill to climb toward recovery in the event of a disaster.

The collaborative objective of the Sendai Framework for building resilience and reducing risk coincides with "*a renewed sense of urgency within the context of sustainable development and poverty eradication*", and to integrate both DRR and the building of resilience into policies, programmes and budgets at all levels (SFDRR, 2015). Moreover, it intends to guide the efforts of stakeholders at all levels: global, regional, national and local. Commenting on the impact that the COVID-19 pandemic will have on resilience, Djalante et al highlight the importance of early recovery planning that is centred around inclusivity. Moreover, the authors emphasize the benefit of DRM and CCA integration on reducing the health burden resulting from climate change and disasters (Djalante et al 2020; 5). This recognition is not a new realisation amidst the current global backdrop but has been highlighted in recent years. According to Banwell et al (2018), an increase in well-planned, effective and appropriate adaptation and risk-reduction is necessary over the short-, medium- and long-term to address climate change and climate-sensitive disaster risks, once again placing disaster preparedness as integral in disaster risk management.

#### 2.7 Resilience and Vulnerability in CCA & DRM in Irish Frameworks

The Emergency Management Structures and climate change adaptation measures in Ireland often use the concept of 'resilience' interchangeably throughout policy formation. In the National Adaptation Framework (2018), it is acknowledged that the concept of resilience is present within both policy areas, and that the concept could assist in integration by providing *"common ground upon which more coherent policies and actions might be built"* (NAF, 2018; 34). Despite not specifically referencing the Sendai Framework, Ireland does highlight resilience as a key element of risk management, and *"achieving the optimal outcomes from the response effort when an emergency occurs"* (SEM, 2017).

The concepts of resilience and "community spirit" are referenced throughout the SEM Framework. However, writing from an international perspective, Siders (2020) raises an important question, namely what exactly does the concept of resilience mean, and how can ambiguous understandings of resilience increase the effectiveness of efforts to build a better future. As a key concept at the intersection of both fields, it is important to note that – if not clearly defined – the word can even have negative implications in disaster risk management.

# 3. Challenges, Gaps and Opportunities in Ireland

#### 3.1 Examples of Good Practices

Ireland's progress in both CCA and DRM has greatly strengthened in recent years, with the adoption and implementation of the various institutional and policy frameworks discussed, as well as linking increasingly detailed extreme weather projections into sectoral adaptation planning. However, there is much to be gained from improving the alignment and integration of DRM and CCA, reducing risk and increasing resilience. Venton and La Trobe (2008) have observed that DRM measures will lessen climate impacts, and adaptation measures will reduce disaster risk. Moreover, the similar aims and agenda of DRM and CCA highlight the complementarity of both areas, and the benefits of increasing awareness and knowledge development across both topics. The following section will highlight integrated CCA and DRM measures currently in place in Ireland that address important climateinduced hazards on a sectoral level.





#### Flooding

Flooding remains the primary focus within national frameworks and plans. Sea level rise, coupled with more frequent and extreme storm events will see as many as 70,000 Irish addresses exposed to damage from coastal flooding, according to a recently published White Paper by the Gamma Location

Intelligence data company (Cantwell, 2020). Flood risks are no longer only considered a concern in 'hot spots' along rivers and coastlines, with surface water flooding now a growing concern in many areas (Mehryar & Surminski, 2020). In the face of such threats, forward looking risk reduction is important in building resilience of individuals, business and governments (ibid). An increase in more intensive rainfall patterns have meant that local authorities, as the lead state body in preparation and response to flood risk, are required to undertake more rigorous flood resilience planning while also considering a range of adaptation efforts (Clarke & O'Donoghue-Hynes, 2020; 45). Figure 7 above presents the relevant bodies involved in flood risk management in Ireland.

The Climate Change Sectoral Adaptation Plan for Flood Risk Management was produced in 2019 under the National Adaptation Framework (2018). It outlines the potential impact of climate change on flooding and risk management in Ireland and identifies key objectives and sustainable approaches to adaptation as part of flood risk management (OPW, 2019). The long-term goal adopted by the Plan is *"Promoting sustainable communities and supporting our environment though the effective management of the potential impacts of climate change on flooding and flood risk"* (OPW, 2019; 2). The OPW has identified 3 objectives to achieve this aim outlined in Box 1 below:

#### Box 1: OPW Objectives on climate change and flood risk (OPW, 2019; 2)

**Objective 1:** Enhancing our knowledge and understanding of the potential impacts of climate change for flooding and flood risk management through research and assessment.

**Objective 2:** Adapting flood risk management practice to effectively manage the potential impact of climate change on future flood risk.

**Objective 3:** Objectives to pursue aligning adaptation to the impact of climate change on flood risk and flood risk management across sectors and wider Government policy.

The cross-sectoral impact of emergencies, referenced in the previous section, is reflected in Objective 3 of the Climate Change Sectoral Adaptation Plan. The interdependency of all sectors is strengthening and growing more important given that these extreme weather events can result in a cascade of impacts on energy, transport and health infrastructure (Leitner, 2020). Aligning adaptation plans with other sectors and across Government policy can significantly help to foster collaboration to prevent and prepare for potential disasters, meet the challenges of the current century and a noteworthy practice for potential replication in other sectors. However, it is important to note that the plan itself does not specify how other sectors should provide for potential climate-related changes in flood hazard and risk, "as it is relevant to their assets and/or areas of interest" (OPW, 2019; 23).

#### The CFRAM Programme

As set out in EEA Report (no.15/2017), in many instances, CCA and DRM are dealt with jointly, but may not be labelled as such. An example of this is Ireland's Catchment-based Flood Risk Assessment and Management (CFRAM) Programme. The CFRAM Programme conducted an extensive and detailed analysis to fully assess the risk of flooding in urban and coastal areas. Completed in 2018, the objectives of the programme were to:

- 1. Identify and map existing and potential flood hazard and flood risk in areas at significant risk from flooding and calling Areas for Further Assessment (AFA).
- 2. Identify feasible measures to effectively manage risk in the AFA's.
- 3. Prepare a set of flood risk management plans that set out feasible measures and actions to manage flood risk in these areas (OPW, 2019; 16).

Through the CFRAM Programme, the Office of Public Works (OPW) has assessed and mapped the flood hazards and risk for communities from frequent, minor flood events up to very rare and extreme flooding – providing insightful information for the OPW, local authorities and other sectors in effective adaptation planning against increased flooding risks (OPW, 2019). Moreover, the programme coordinates Ireland's 'whole of Government' approach, informing future planning and helping communities plan for and respond to a flood event. The Flood Risk Management Adaptation Plan expresses that the work undertaken by the OPW – through the CFRAM Programme – *"contributes towards a number of priorities identified under the Disaster Risk Reduction Framework"* (OPW, 2019; 18-19), namely Priority 1, 'Understanding disaster risk', Priority 3, 'Investing in disaster risk reduction for resilience', and Priority 4, 'Enhancing disaster preparedness for effective response, and to 'Build Back Better' in recovery, rehabilitation and reconstruction (UNDRR, 2015).

#### Nature-Based Flood Relief Scheme

Nature-based solutions have been identified as another potential avenue to enhance integration of CCA and DRM while simultaneously boosting community resilience (Natoli, 2019). Dublin City Council implemented its Environmental Improvement Scheme – consisting of realignment of footpaths and road surfacing works – while also taking into account flood risk. The design included the management of surface water runoff in a sustainable way, rather than the conventional piped system. This is intended to increase the capacity to deal with extreme rainfall events in an urban setting, whilst contributing to flood resilience by removing surface water discharges from the combined system. A nature-based 'engineered' solution to manage surface water from the preliminary design stage was a cost-effective and climate resilient development, providing better flood management and climate resilience (Clarke & O'Donoghue-Hynes, 2020; 103).

#### Infrastructure and Transport

The transport sector in Ireland is an example of sectoral climate adaptation, which incorporates strategies for various weather-related events that will have an impact on the national road and light rail network. Transport is used by the majority of the public and is therefore fundamental to the functioning of society. Under a changing climate, meteorological and climate parameters can develop in uncertain ways, and make consequences for transport networks much more difficult to predict (ITF, 2016). The Irish road network is particularly vulnerable to both direct and indirect impacts of climate change. Flooding directly impacts the land transportation service, while degradation of the road service and loss in structural integrity indirectly induces higher maintenance and repair costs. Similarly, for the rail network, inundation of rail sections or damage to railway stations is a high possibility with projected changes in rainfall patterns (Hawchar et al, 2019). An example of this is the Dundalk-Dublin rail line, which is forecasted to see a significant increase in rainfall by mid-century (ibid). Having a coherent framework for adaptation can ensure that risks are outlined, an allocation of responsibilities is undertaken, certain events are prioritised, and strategic decisions are not neglected or overlooked (ITF, 2016).

The Transport Climate Change Adaptation Sectoral Plan was implemented in 2019 under the NAF. In the strategy, the Sendai principle of 'Build Back Better' is integrated throughout. The report notes: *"When infrastructure is damaged during an extreme weather event, consideration should be given to not only restoring operations but also to whether newer technologies or amended capacity is required to future proof the asset"* (DTTS 2019; 14). Moreover, the role of the transport sector in emergency planning was highlighted, with the recognition of climate change adaptation as a co-benefit to response plans in minimising the impact of extreme weather events (ibid; 14). The implementation objectives set out in this plan are:

- 1. Improve understanding of the impacts of climate change on transport infrastructure, including cross-sectoral cascading impacts, and close knowledge gaps;
- Assist transport stakeholders in identifying and prioritising climate risks to existing and planned infrastructural assets and enabling them to implement adaptation measures accordingly; and
- 3. Ensure that resilience to weather extremes and longer-term adaptation needs are considered in investment programmes for planned future transport infrastructure (DTTS, 2019; 84).

In order to achieve these aims, the plan additionally proposes a number of comprehensive adaptation actions, identifying lead departments and stakeholders, and clear linkages between sectors.

Future climatic factors have been factored into road schemes, according to Transport Infrastructure Ireland (2017). In its report on adapting to climate change, it provides an overview on the main challenges it faces, while conveying its strategy for managing, improving, preventing and cooperation with climate-related weather events that impede on transport infrastructure (TII, 2017). According to the report: *"Every effort is made during the planning phase to reduce the impact of flood events, and where feasible, the roads are raised above flood plains"* (TII, 2017; 5). A number of preventative measures have additionally been put in place to minimise the impact in the event of disruption, including identifying hot spots and preparing action plans, and ensuring that future schemes take climate change into account in planning and construction (TII, 2017; 13). Despite not referencing the Sendai Framework in the SEM Framework (2017), the Strategic Emergency Management Guideline 3 on Critical Infrastructure recognises the concept of 'Build Back Better' in regard to investment, and outlines *"It should be the aspiration of all owners/operators to use an opportunity, within financial constraints, to replace or expand CI with new/robust/technologies which have greater resilience to envisaged threats " (DoD, 2019; 22). Moreover, it identifies that there is a cost-benefit analysis in adding robustness to critical infrastructure (ibid).* 

#### Health

Health risks related to climate change are of growing concern worldwide, with the impacts of climate change not only exacerbating existing health problems, but also the threat of new and emerging health issues (UNFCCC, 2017). A Synthesis Report conducted by the Secretariat of the UNFCCC found that certain groups will face disproportionate health impacts from climate change as a result of age, gender and marginalisation. Additionally, many vector-borne diseases are highly sensitive to climate conditions; for example, by lengthening the transmission season and expanding geographical range (UNFCCC, 2017). The importance of reducing climate vulnerability in the health sector in Ireland has been underpinned in the Department of Health's first Climate Change Sectoral Adaptation Plan (DoH, 2019). The plan was prepared under the National Adaptation Framework and pays special attention to particular health vulnerabilities and vulnerability of communities in Ireland. A detailed risk assessment on the health implications of climate change - informed by the MEM approach considered the likelihood of a climate scenario or event based on climate projections, as well as the potential direct and indirect health impacts that may have on communities in Ireland (DoH, 2019). The Health Service Executive (HSE) operates within across all four Major Emergency Management Regions, each region with its own Emergency Management Plan (HSE, 2020). In 2016, a study was conducted on Irish nurses' preparedness for an influenza pandemic. The study indicated that preparation and planning were "crucial in maintaining essential services and the integrity of the healthcare system during an Influenza pandemic" (McMullan & Brown, 2016; 11). This is an example of the necessity of planning and preparedness in disaster risk management, acting as the threshold between mild and severe emergencies. Health is considered an important endpoint for both

climate change and disasters, with reducing vulnerability incremental to addressing the health impacts of CCA and DRM (Banwell et al, 2018; 2). Health sector preparedness has been thoroughly tested by the COVID-19 pandemic and will no doubt be the subject of considerable research attention and lessons learnt in the coming months.

#### 3.2 Challenges and Barriers in Integration

The National Adaptation Framework (NAF, 2018) highlights the growing need for integration of emergency planning, in particular disaster risk reduction, and climate change adaptation. The Framework recognises the positive ramifications of responding to short-term challenges while building long-term resilience. Moreover, the NAF acknowledges the importance of integration between DRM & CCA policies so that *"potential co-benefits are identified and exploited"* (NAF, 2018). Despite originating from different knowledge backgrounds, the converging aims of CCA and DRM should allow for greater impacts through law and policies; more efficient use of available resources; and more effective action in reducing vulnerabilities (Natoli, 2019).

In Ireland's Adaptation Preparedness Scoreboard for the European Commission, it was stated that mechanisms to integrate and coordinate CCA and DRM approaches were "in progress", and that Ireland would shortly publish National Framework for Emergency Planning – which will "include a guideline on climate change adaptation" (EEA, 2017; 15). The SEM Framework (2017) contains no explicit reference to climate change adaptation at any point, despite Government recognition of the common interests both paradigms share. Additionally, the Sendai Monitor - similar to the EU Adaptation Preparedness Scoreboard - monitors and reports achievements in global targets for the SFDRR. In the years 2015 and 2016, the Government reported on all seven of its' Sendai targets. In 2017 and 2018, Ireland only reported on some of the Sendai targets. Last year, the Sendai Monitor did not receive any update from Ireland (UNDRR, 2020). A set of 38 indicators were established to monitor progress on the Sendai Global Targets. Indicator E1 advises that "policy coherence relevant to DRR such as sustainable development, poverty eradication and climate change, notably with the SDG's and the Paris Agreement" (UNDRR, 2019; 253). Despite Ireland's recognition in the National Adaptation Framework (2018) that complementarity between the Sendai Framework and other international instruments such as the SDGs and the Paris Agreement is necessary, the lack of crossreferences to these respective international frameworks in the relevant Irish policy documents may be a missed opportunity.

There is much to be gained from improving the alignment and integration of DRM and CCA, as a means of reducing risk and increasing resilience. Venton & La Trobe (2008) have observed that DRM measures will lessen climate impacts, and adaptation measures will reduce disaster risk. Moreover,

the similar aims and agenda of DRM and CCA would benefit both areas, increasing awareness and knowledge development in both paradigms. According to the OECD's most recent Development Cooperation Peer Review (2020), Ireland places a great focus on climate change adaptation in its international development co-operation through Irish Aid. However, the review notes that *"Ireland also needs to improve the mechanisms it uses to identify potential areas of incoherence between its domestic policies and development objectives and to systematically assess and monitor progress in addressing these"* (OECD, 2020; 18). The following section will aim to highlight the various barriers and challenges identified in relevant literature that prevent integration of DRM and CCA. Moreover, it will identify the gaps and opportunities that can promote integration of both paradigms.

#### Institutional Arrangements

The ESPREssO Project observed that in many instances, on an institutional level, decision-making and policies materialise from different sectors (Amaratunga, 2017; 40).

"Institutional barriers are identified as a key challenge that hinders the process of successful integration of CCA into DRR ... In most of the countries climate change related policies and decisions are made by the ministries and organisations related to the environment, whereas disaster management and reduction decisions are made by ministries related to infrastructure development. This institutional structure **disturbs the communication process** which generates an information barrier among the institutions" (Amaratunga et al, 2017; 40).

The institutions that deal with disaster response and recovery, disaster risk reduction and climate change adaptation are typically separate. In Ireland, the principal body in charge of adaptation policy-making lies within the Department of Communications, Climate Action and the Environment (DCCAE), with CARO regions coordinating climate change adaptation at a local level. In disaster response, responsibilities and arrangements vary across a range of stakeholders, with PRA's assuming the majority of responsibilities in emergencies, alongside varying LGD's, depending on the particular emergency. Monitoring and reporting are completed internally by LGD's.

In terms of vertical alignment between the international and national levels, the Sendai Framework, the key global instrument on disaster risk reduction, is not referenced within the SEM Framework, and although the SEM Framework integrates the four priority actions of the Sendai guidelines, express references to international instruments would allow for more coherent and integrated policies. Likewise, the Strategic Emergency Management Framework (2017) has a comprehensive approach to response and recovery at its core. However, its strategy for focusing on planning, preparation and capacity for adaptation may be not as extensive, despite a dominant focus on risk reduction at an international and EU level, limiting the potential scope of integrating DRM and CCA. A statement by

31

Ireland at the Third World Conference on Disaster Risk Reduction: "*A common international approach is more effective than separate national approaches. Disasters do not respect borders*" (Government of Ireland, 2015). It follows that a vertically aligned approach would likely be more effective with a clear linking of emergency management and disaster response measures from an international – to national – to local level.

#### **Unequal Funding**

Funding has been considered a major barrier in the alignment of CCA and DRM strategies. While integrated approaches are largely recognised as mutually beneficial, a challenge remains in integrating both approaches. Sandholz et al (2020) indicate that little is known about the actual types and magnitude of the cost of policy incoherence. Moreover, according to the PLACARD Project, finance and funding of CCA and DRM may come from separate funding systems, thus creating difficulty in integrating the two sectors.

In relation to disaster management, Ireland, as per the SEM Framework, predominantly addresses response and recovery, with a focus on *"restoring normal functioning"* (DoD, 2017; 28). This is perhaps due to the fact that adaptation measures for critical infrastructure are typically associated with high costs. In many instances, the best (or most cost-effective) option may be not to implement adaptation measures (DTTS, 2019). Since 2009, the DHPLG has made financial support available to assist local authorities in meeting the cost of response and recovery in severe weather events (DHPLG, 2019). This assistance is provided on an ad-hoc basis from savings identified elsewhere in the department budget. The LGD of the particular emergency is to collate costs associated with emergency events (DHPLG, 2019); however it is not specified whether additional funding is provided for increasing the resilience of infrastructure or services against future climate-induced disasters as per Sendai priority 4 to 'build back better'.

However, there is emerging literature to suggest that a lack of integration of the two areas could lead to great financial repercussions (IFRC, 2019). In Ireland, severe flooding that occurred in Galway during Storm Desmond in 2015 generated an estimated cost of  $\in$ 3.8 million for the road network alone (Clarke, 2019; 2). Moreover, it has been estimated that the road networks in Ireland and in the UK could experience cost increases from  $\in$ 59- $\in$ 89 million per year over the period 2070-2100 (ibid), highlighting that in order to develop adequate emergency planning and response measures, it is integral to quantify the impact of climate change. This is reflected in the latest National Risk Assessment overview, where it is acknowledged that "*the cost of inaction exceeds the cost of action, and that this differential will rise steeply with time*" (Government of Ireland, 2019; 15).

The transport sector, in its adaptation plan, noted the absence of a standardised framework to associate climate or adaptation specific actions with financial investment. The plan identified this as a major knowledge gap, and a barrier in understanding climate resilience within organisations and structures across all sectors in Ireland (DTTS, 2019; 80). Effective disaster risk reduction measures are considered a particularly effective way to limit damage and fatalities when compared to response and recovery, according to a study conducted on the cost-efficient nature of DRR (Hugenbusch & Neumann, 2016).

#### Misaligned Time Scales

Another point of friction when attempting to integrate CCA and DRM priorities at the national level is the conflicting time scales. Climate change adaptation efforts tend to focus on long-term projections and predicted effects, thus with a heavy reliance on science-based perspectives, whereas members of the DRM sector predominantly work in the humanitarian sector, and focus more on learning from past experience, with more of an emphasis on localised needs (Natoli, 2019; 11). This variation in timescale presents a situation where the immediate impacts tend to overwhelm the capabilities of the affected population and rapid responses are required (Thomalla, 2006). Nonetheless, these inconsistencies in timescales have reduced over the last number of years, through the efforts of CCA and DRM communities in aligning perspectives and actions. Ireland's National Overview of Strategic Risks provides an overview each year of risks considering the "next twelve months" (Government of Ireland, 2019; 8). The plan highlights the threat of climate change, with the most "immediate risks" considered to be changes in extremes, floods, precipitation and storms (Government of Ireland, 2019; 51). In the National Risk Assessment, completed every five years under the EU Civil Protection Mechanism (2013), it stipulates that consideration was given to the "potential impact of climate change" (DoD, 2017; 24); however these potential impacts were not specified in the context of each risk, or throughout the assessment. A time horizon of 12 months for an overview of strategic risks, and a 5-year time span for the National Risk Assessment – according to the EEA – is "too short" to inform long term investments in climate risks (EEA, no.15/2017; 126).

#### Knowledge Gap

The resilience of a community is inextricably linked to the condition of one's physical, social and natural environment. With a growing emphasis on disaster resilience – there has emerged a better understanding of the concept of vulnerability, with a collective realisation of the concepts of resilience and vulnerability being closely related, each contributing to the understanding of disaster impacts and consequences (Tierney, 2014; 165). The ESPREssO Project noted a significant lack of priority dedicated to vulnerability assessments at a national, regional and local level within Europe. According

to Zuccarro et al (2018; 872), there is an obligation for complex and multidimensional risk assessments that consider social and economic vulnerabilities.

On a national level, Ireland's international development and cooperation policy, *A Better World*, largely focuses on "the furthest behind first", calling for greater investment in conflict prevention and tackling root causes of vulnerabilities (Government of Ireland, 2019; 2). Ireland has pledged at least  $\in$ 10 million per year for the next three years to the UN Emergency Response Fund (Government of Ireland, 2019b; 17), as well as to "build the resilience of the poor and those in vulnerable situations and reduce their exposure and vulnerability to climate-related extreme events and other economic, social and environmental shocks and disasters" (Government of Ireland, 2018, 222). However, there is scope for domestic progress, with the National Adaptation Framework (2018) acknowledging that further research is needed on vulnerabilities in Ireland.

Ireland's risk assessments, the National Risk Assessment (2017) and Overview of Strategic Risk (2019), both refer to vulnerability in the context of infrastructure, without acknowledging in-depth the human implications of vulnerability and risk. Likewise in the SEM Framework, in reference to critical infrastructure, it is expected that any disruptions caused "will, as far as possible, be managed in an equitable manner to minimise the impact on society as a whole and in particular vulnerable people" (DoD, 2017; 19). Similarly, the National Adaptation Framework explicitly references the concept of vulnerability throughout, noting that "climate change is likely to disproportionately impact on the lowest socio-economic groups in society", and that these groups are also the "worst positioned to adapt to the changing climate" (NAF, 2018; 20). However, it is not specified in any framework who is acknowledged as a vulnerable group, and whose needs should be prioritised in the event of a disaster. Natural hazard impacts are unevenly distributed, and Thomalla (2006) notes that in order to reduce vulnerability, there must be a clear understanding of who is vulnerable to the impacts, and how the interactions between nature and society shape the fundamental processes that contribute to vulnerability. Despite the Major Emergency Management Framework highlighting the needs of vulnerable groups in communities when assessing risk, the SEM Framework appears to be onedimensional in its approach, with no evaluation on the human impact of climate change on vulnerable or at-risk groups. Considering the MEM Framework dates back to 2006, the context of risk and vulnerability it refers to may be outdated, particularly in the context of the increasing risks posed by climate change.

As stipulated by Albris et al (2020), disasters have social roots, and are equally related to societal conditions as they are to hazard exposure and geographical location. While disasters take place at a point in time, it is the societal and economic losses that elongate these repercussions (Tierney, 2014; 125). There are various groups in Ireland that have been and will be disproportionately impacted as a

result of a disaster. If this is not highlighted within risk assessments and implementation plans, Ireland may be susceptible of overlooking the needs of many vulnerable groups. Despite EU Member States making efforts to improve the coherence and transparency of risk assessments at a national level, Zuccaro (2018) suggests there is an absence of a common methodology and differing risk/impact variables (hazard, exposure, vulnerability and impact on communities) (Zuccaro, 2018; 12).

#### Language Barrier

As discussed by Siders (2020), if there are inconsistencies in terminology or fragmented information, this can lead to potential differences in understandings, further separating both communities working on DRM and CCA. The National Adaptation Framework identifies the term 'resilience' as a common ground to build more integrated policies and actions upon (DCCAE, 2018). However, it is only an opportunity for integration if the definitions used in both frameworks are cognisant of one another. Although differences in terminologies are often valid, separate understandings of the concept of resilience may indicate discrepancies in the way both communities consider vulnerability (Barrott & Bharwani, 2017).

The MEM Framework (2006) provides guidelines on carrying out risk assessments in which it outlines a four-stage approach to determining risk at a local and regional level: i) establishing the context; ii) hazard identification; iii) risk assessment; and iv) recording hazards in a risk matrix (DHPLG, 2006). Each of these approaches require measurement of a societal impact of an emergency. In a disaster risk context on a national level, the term resilience in the MEM Framework (2006) is defined as "the term used to describe the inherent capacity of communities, services and structures to withstand the consequences of an incident, and to recover/restore normality" (OPW, 2006; 29). In the SEM Framework, although the term resilience is not defined, *community resilience* is considered as "the sustained ability of a community to mobilise available resources to respond to, cope with, and then recover from adverse situations, such as emergencies" (DoD, 2017; 18). Both definitions describe resilience as a way of returning to normal, or the ability to withstand an incident, or 'bounce back' (Siders, 2020; 117). In contrast, the concept used in the National Adaptation Framework is of 'climate resilience' which is defined as "the capacity of a socio-ecological system to absorb stresses and maintain function in the face of external stresses imposed by climate change and adapt, reorganise and evolve into configurations that improve the sustainability of the system, leaving it better prepared for future climate change impacts" (DCCAE, 2018; 61). With this definition, there is more of an emphasis on evolving, progression and 'bouncing forward' (Siders; 2020; 119), rather than returning to pre-disaster conditions. When practitioners do not consider their actions outside the scope of their field, they fail to consider the concept may have an entirely different meaning. Both CCA and DRM originate from different backgrounds, with decision-makers in separate sectors arriving at different solutions and – as Siders (2020) highlights – with many of the consequences unknown.

Moreover, in many cases, it is often not an issue of *whether* there are integrative practices, but *where* and *how* these are located within the relevant frameworks. As the EEA (2017) highlights, these good practices are often not explicitly referenced as integrative practices, but due to the nature of the practice/policy, there is both a climate change adaptation and disaster risk element. This will be discussed further in the following section.

#### Strengthening Participation

From this initial review of the literature, there is a potential challenge in Ireland regarding the systematic incorporation of community perspectives from vulnerable persons or sectors into risk management planning and risk reduction. There is already a great level of community engagement and public consultation in raising awareness on climate change and the increase of frequency and magnitude of severe weather events, as conveyed in the level of detail and community engagement which preceded the development of many of the Local Authority Climate Action / Adaptation Strategies. Having said that, a stronger focus on vulnerabilities would enable a more inclusive risk planning process. Mayo County Council in its Adaptation Strategy notes the need to identify communities which are vulnerable to the impacts of climate change and extreme events, and the need to develop resilience plans, but flags that these activities are not budgeted (Mayo County Council, 2019; 102). Similarly, Cork County Council acknowledges plans to develop a programme to enhance capacity to respond and recover from extreme weather events: "with specific aims to help the vulnerable community to develop a stronger facilitating role" (Cork County Council, 2020; 56); however this particular action it is not included within the budget. The OECD – in its most recent Development Cooperation Peer Review – noted that Ireland still lacks a clearly articulated approach to private sector engagement, particularly in the agri-food sector (OECD, 2020). The report notes that the private sector strategy that is currently being created should "have a clear focus on the development impact of Ireland's private sector engagement on marginalised populations" (OECD, 2020; 19). The concept of 'just transition' and a 'just recovery' have taken on a new meaning in recent months. As the completion of this literature review takes place against the sombre backdrop of a global pandemic, the relevance of emergency planning and preparedness is reflected globally. As Ireland recovers socially and economically from the COVID-19 period, the promotion of a 'just recovery' - that is inclusive and prioritises the vulnerable, as well as recognising the needs of all (Djalante, 2020) - is at the very core of why CCA & DRM integration is so relevant. Both are equally important processes on managing future risks and galvanising cooperation between communities, across sectors and beyond borders (Stockholm Environmental Institute, 2020). Focusing on building long-term resilience while responding to short-term concerns is paramount.

## 4. CCA & DRM – Bridging the Gaps

The present research has noted many examples of integrated good practices located within the literature, as described in the previous section. These examples, although not explicitly referenced as such, have the potential to act as a benchmark for more integrated and cohesive policies. The following section will explore further potential pathways for enhancing integration.

#### 4.1 Risk and Adaptation Financing

Investment in disaster risk reduction and prevention is essential to enhance economic, social, health and cultural resilience of persons. Resources for integrated funding could derive from budgets, international actors and private donors. In the 2020 Budget, the Minister for Finance identified climate change as one of the greatest long-term challenges facing Ireland and announced an increase in the price of carbon from  $\notin 20$  per tonne to  $\notin 80$  per tonne by the year 2030 (Oireachtas TV, 2019). This measure is aimed to raise €90 million in 2020 – which will be ringfenced to fund new climate measures. Moreover, a dedicated 'just transition fund' will be created to address priorities, determined by local communities (Oireachtas TV, 2019). The National Development Plan (NDP) 2018-2027 sets out investment priorities that underpin implementation of the National Planning Framework: Project Ireland 2040 (2018). Speaking on the plan, then Minister for Climate Action and Environment Denis Naughten said: "Project Ireland 2040 represents a significant step-change in our approach, both in scale of our ambition and funding that we're making available to meet the challenges. Well over €1 in every €5 is to be spent under the NDP on climate mitigation and adaptation" (O'Sullivan, 2018), with adaptation measures financed in the agri-food sector, land use, forestry and flood risk management (Government of Ireland, 2018). In response to weather emergencies, councils spent a total of €101 million between 2014 and 2018 on areas such as clean-up, road repair and disaster recovery (LGMA, 2020).

Ensuring risks are understood as part of investment decisions, and financing for disaster preparation and recovery takes place in an equitable way, will encourage the integration of CCA and DRM into emergency planning. A potential avenue for integration is to allocate funding where resources are better leveraged to achieve the twin objectives of both sectors. For example, investing in Pre-Disaster Recovery Planning (PDRP) enables effective coordination and decision-making structures, facilitating swift and informed actions. Moreover, it increases the likelihood that planning processes are more inclusive of all stakeholders, including vulnerable groups that perhaps would not have the capacity to participate amidst a disaster (UNDRR, 2017). Additional resilience could be strengthened through appropriate social protection and schemes aimed at reducing vulnerability (Natoli, 2019). Improved risk data, according to UNDRR (2017), will lead to more efficient and stable financial markets, thus contributing to more effective investment. Promoting consistent use of accurate and appropriate disaster risk and climate risk scenarios by governments, public institutions and financial regulators, as well as private sector firms will explore how existing data can be used to support investors and citizens (UNDRR, 2017). Forecast-based financing is an example of targeted disaster risk financing, which includes anticipating extreme weather events to implement actions before the impact is experienced. Therefore, the focus is more on mitigation, preparedness and emergency response. Forecast-based financing provides assistance for the most vulnerable and at-risk groups before a potential disaster (Leitner et al, 2020).

The transport sector acknowledges the importance of Building Back Better in their climate change adaptation sectoral plan (DTTS, 2019). However, as these principles require a whole of Government approach, for instance, in transport-appropriate land use, building and planning regulations are employed to reduce the possibility of building infrastructure in high risk areas. Consequently, it may be difficult to ensure all sectors are employing these principles if they are not explicitly within national policies and frameworks. Employing 'Build Back Better' priorities will ensure that recovery process can be leveraged to increase resilience of infrastructure at the point of (re)construction. When infrastructure is damaged during an extreme weather event, consideration should not only be given to restoring infrastructure, but considering whether future technologies or more robust groundwork is required to future-proof the asset from eventual climate-related weather events.

#### 4.2 Cross-sectoral Communication, Coordination & Collaboration

In a recently published PLACARD report on strengthening the two approaches, it was identified that in ensuring sound governance in integrating CCA into disaster risk management, there were often challenges faced, such as separate user and stakeholder engagement processes, particularly when many targets of the two spheres were similar (Leitner et al, 2020; 8). According to the ESPREssO Report, a strengthening the interconnectedness of policy response to CCA and DRM, such as better collaboration among stakeholders on a multilateral level, coupled with shared indicators, monitoring protocols and concepts would foster a synergy between both areas, leading to greater social, economic and environmental outcomes (Zuccaro, 2018). A challenge identified by Torney (2020) on the sectoral and local climate change adaptation plans relates to their *"relative newness"* (ibid, 1). Although there is a stronger governance structure than previously existed, requiring sectors to develop CCA plans in isolation does not encourage the collaboration and coordination needed for effective climate adaptation. The author suggests a potential designated entity to *"facilitate collaboration across sectors to develop one national adaptation plan"* (Torney, 2020; 2).

Although CCA and DRM both pursue complimentary aims, they have different structures and policies, thus possessing different norms and value systems. Such differences will not be connected by legislation and financial incentives alone. Communication, coordination and collaboration is key in overcoming the incongruity that lies between both climate change adaptation and disaster risk management. Approaches aimed at strengthening horizontal coordination and identifying roles and liabilities of key stakeholders across all levels, as well as "*streamlining 'top down' and 'bottom up' policies towards participatory models that go beyond mere risk awareness and communication issues, to be focused on community engagement mechanisms able to operationalize knowledge-sharing across relevant sectors"* (Zuccaro, 2018a; 12). In preparation for the Flood Risk Management Climate Adaptation Plan, a consultation workshop was held for stakeholder groups, including local authorities, government departments, members of academia, and operators of critical infrastructure, with findings from the workshop informing the preparation of the Adaptation Plan (OPW, 2019). Moreover, in the drafting of the flood risk management sectoral adaptation strategy, a meeting was held between the OPW and CARO representatives, resulting in a comprehensive adaptation policy that assessed risk.

The series of extreme weather events which hit Ireland during the period 2016-18, including exhurricane Ophelia, Storm Emma and the heat wave and drought of 2018, led to a detailed analysis and lessons learnt report conducted by the National Directorate for Fire and Emergency Management in the Department of Housing, Planning and Local Government (2019). The report identified a strength in the communication between local government and the DHPLG, with well-established relationships and effective communication resulting in a clear identification of what needed to be addressed, and coordinating arrangements based on those needs (DHPLG, 2019).

#### Local Level Integration

Integrated approaches at a local level, would identify various vulnerability factors, contributing to successful implementation, according to Cutter and Osman-Elasha (2018). Under the Local Authority Adaptation Strategy Development Guidelines, local authorities are expected to identify, monitor and assess climate hazards, impact and consequences to increase capacity to plan accordingly for adaptation (DCCAE, 2018). A questionnaire conducted by Clarke & O'Donoghue-Hynes revealed that *all* local authorities had identified or were in the process of identifying the full range of weather events likely to affect their areas, with significant progress in identifying key actors involved. Moreover, almost all local authorities had made significant progress in assessing and documenting the potential future risks posed by extreme weather and climate variability (Clarke & O'Donoghue-Hynes, 2020). Local infrastructure greatly varies among communities, and when plans are tailored to a specific municipality – with the community contributing to the plan – that local knowledge will be of great benefit in identifying risk and resilience factors. This community-based approach is considered an integral element in identifying priorities and objectives when considering climate-smart

39

disaster management, with particular regard to vulnerable and marginalised groups. Moreover, it would aid in the development of inclusive and sustainable solutions and community resilience potential (Natoli, 2019). Cork County Council, in its adaptation plan, identifies specific targets to support the successful and practiced implementation of adaptation planning, such as mainstreaming CCA into plans, programmes and policies, including the severe weather plan (Cork County Council, 2019). Clarke & O'Donoghue-Hynes (2020) highlight that continuous and comprehensive planning could facilitate local authorities to access and coordinate responses to extreme weather events and climate variability in a "*systematic, evidence informed and consistent manner*" (ibid, 2020; 26).

However, it must also be highlighted that the availability of local data may in fact be limited. Moreover, as illustrated by Leitner et al (2019), there is a lack of Sendai-aligned strategies at a national level, impeding comprehensive CCA and DRM integration at a local level. As previously mentioned, Ireland's SEM Framework, although an integral plan for building resilience of communities in response to disasters, it is not explicitly Sendai-focused, which limits its scope in terms of vertical alignment. Moreover, in a report conducted by Clarke & O'Donoghue-Hynes (2020), it was established that many projects conducted by local authorities run independently of one another in addressing certain issues, thus "losing out on the opportunity to share their learning or partner and create synergies around projects that were relevant to other local authorities" (ibid, 8). Sharing knowledge and expertise in the paradigms of CCA and DRM can be greatly beneficial, not only from an international to local level, but from the local to the global, with a reciprocal exchange monitoring actions taken at each level (Cubie & Natoli, forthcoming). As Zuccaro (2018b) suggests, further reflection should examine how to develop harmonised protocols and procedures, and a standardisation of assessments among countries to enhance coherence. While acknowledging that it should not be a one-size fits all model and that effective integration is often country and context specific (Natoli, 2019; 12), it is important to provide a basis for sharing experiences, implementation of common risk strategies and cross-national validation (Zuccaro, 2018b; 17).

A potential point of connection could exist between the CARO and MEM Regions. CARO's are based on regions which face similar risks to climate change (Cork County Council, 2019), such as sea level rise, coastal flooding, storms, ground water flooding and rural pluvial flooding. These risks require effective emergency management. The MEM Regions are grouped into eight regions which have been created for major emergency purposes (Cork County Council, 2019). Looking at the SEM and MEM structures through a climate change adaptation lens could mean more effective and stringent policies, through more effective use of resources, and improved prevention and preparedness to climate risk. Likewise, the viewing of the National Adaptation Framework and Climate Action Plan through a disaster risk reduction lens would allow both paradigms to work in tandem, reinforcing one another. An adaptation plan consisting of actions that foster collaboration across departments and with other stakeholders, will require tools to support synergistic development. Well-designed workshops could be an opportunity to discuss issues of concern and possible resolutions (Torney, 2020).

#### 4.3 Envisioning a Harmonised Language for CCA & DRM in Ireland

DRM & CCA are both cross-cutting sectoral issues, and thus it is integral that there is a broad involvement of stakeholders for sufficient cooperation. However, when a number of stakeholders are present, it may become difficult to find commonalities in language and terms used, creating diversified knowledge of what constitutes risk, resilience and vulnerability; all major issues within both DRM & CCA frameworks. Integration of both paradigms leads to better access to knowledge, resources and skills, leading to more coherent and harmonised policies in both sectors, and reducing potential for maladaptation.

As previously discussed, the concept of resilience remains incohesive throughout Irish and international frameworks. Without a clear definition, ambition to integrate both CCA and DRM may lead to further incoherence, due to the contrasting definitions in both paradigms. Further clarity on terminology in both frameworks - such as the concepts of resilience and vulnerability - would increase coherence and align goals; achieving mutual benefits for both fields (Siders, 2020). A concept discussed by the UNDRR within their Guide to Assessment of Integrated Research on Disaster Risk (AIRDR), is transformative disaster risk management. The AIRDR report considers vulnerability reduction, effective adaptation and higher resilience as pathways toward transformation (UNISDR, 2015). Gotham & Campanella (2010) use the term 'transformative resilience' in the context of an urban ecosystem, and how "indicators, properties and drivers of post trauma urban eco-system transformation" (Gotham & Campanella, 2010; 12). This, in turn, informs how different urban ecosystems can transform and innovate in times of trauma. In this respect, urban ecosystems do not just automatically respond or adjust to trauma but learn from past experiences to reorganise and reinvent themselves in progressive ways. The concept of 'transformative resilience' could be an area of substantial merit from further research, however it falls outwith the scope of the present review. Using language more effectively with shared terminology can be crucial in moving forward toward more integrative practices (Barrott & Bharwani, 2018). Similarly, Ireland would benefit from a ubiquitous definition of vulnerability in all frameworks, ensuring that all stakeholders and relevant key actors are accurately informed. It is worth recognising, in both CCA and DRM fields, that terms can harbour different meanings, so it can be beneficial to search for ways to enhance understandings of how these different terms are used, thereby strengthening knowledge management and increasing opportunities for integration (Barrot & Bharwani, 2018). Placing a dialogue of resilience/vulnerability at the intersection of disaster risk and climate adaptation would provide for a more focused approach in integration of both paradigms.

#### 4.4 Enhancing Knowledge Bases at the Intersection of CCA & DRM in Ireland

The ESPREssO Project identified a lack of platforms for the sharing of knowledge among stakeholders that would enable the progressive application of knowledge in policies concerning DDR and CCA (Amaratunga et al 2017). Based on the research conducted, the Project noted that the limited convergence of science and policy efforts to align CCA and DRM remains a significant barrier (Albris et al, 2020), despite the identified co-benefits for each sector.

Weather-related disasters – including storms, floods and heatwaves – have revealed how vulnerable many Irish sectors, including transport and energy infrastructure, housing stock, agriculture and health, are to climate change (Shine, 2018). A range of vulnerability assessments have been carried out in recent years on the transport, energy and water sectors. However, as stated by Hawchar et al (2019), it is likely that adaptation challenges and opportunities that such sectors face will intersect due to their interdependence on one another. As societies become more complex, in order to effectively manage disaster risk, it is not enough to solely focus on having the means to respond to a particular hazard or event. As expressed by the ESPREssO Project in their Guidelines for Enhancing Risk Management Capabilities: "An advanced understanding of all aspects of a modern, technological society, from electricity systems, urban planning, public risk awareness and their attitudes to such risks, and governmental structures is needed" (Zuccaro et al, 2018; 11). Hawcher et al (2019) suggests that in order to ensure a coherent approach to resilience, a broad vulnerability assessment that considers *all* sectors is necessary (Hawchar et al, 2019). Therefore, responding effectively requires a multitude of stakeholders, and involvement of all parts of society. The recently published PLACARD report recommended the engagement of various stakeholders at different scales, stimulating positive effects in both the decision-making process and the outcomes of fostering coherence between CCA and DRM. The report iterates that the engagement of all concerned actors is critical, "in order to recognise the needs of all" (Leitner et al, 2020; 9).

National risk assessments can play an integral role in developing CCA plans and act as a joint knowledge base for CCA and DRM (EEA, no.15/2017). However, this integration, at a minimum, requires a common understanding and use of relevant risk metrics. Hazard mapping and risk assessment is an area where integration of DRR and CCA is well advanced and recognised as a priority area. According to the EEA (2017), there is still scope for improving coherence between climate change impacts and vulnerability assessment, and the assessment of disaster risk. There is an opportunity to learn from one another, advancing state-of-the-art knowledge that benefits both communities.

## 5. Insights from the literature

The benefits of integrating CCA into disaster risk management in Ireland have been acknowledged by both Governmental and non-Governmental stakeholders. The main policy instruments for climate change adaptation and emergency management incorporate clear language to this effect, and integration is pursued on a local level through adaptation plans, in line with the Local Authority Adaptation Strategic Development Guidelines (2018). On a sectoral level, there is an element of cross-sectoral communication, as highlighted by the OPW (2019). However, the literature suggests that there may be potential for greater vertical alignment with international legal and policy frameworks, such as the Sendai Framework for Disaster Risk Reduction, Paris Agreement and the SDGs. As stated by the Government, a common and unified approach at the global level is much preferred than disparate national approaches (Government of Ireland, 2015). Likewise, at the domestic level, there is a need to ensure the horizontal integration of national laws and policies that impact on climate change and risk reduction. While there is evidence that this is beginning to occur, as seen in the CFRAM Programme, the aim of this research project is to explore examples of good practices across the country as well as working with local and national stakeholders to identify what else needs to be done to promote resilience and reduce vulnerability to the predicted negative impacts of increasing frequency and variability of extreme weather events in Ireland. Issues identified for further exploration include:

- Integration of the concept of build back better during disaster response and recovery, an important aspect of the Sendai Framework, and more explicit linking of Ireland's SEM Framework with the SFDRR priorities for action.
- Unmitigated climate risk may prompt major financial repercussions (Department of Finance, 2018; 28). As noted by the Irish Government, the cost of doing nothing may surmount the cost of action (Government of Ireland, 2019; 15). Mobilisation of financial and economic systems to simultaneously support CCA and DRM efforts would lead to more efficient use of resources (EEA, 2017).
- 3. Mainstreaming integrated DRR and CCA approaches in all sectors, drawing from Irish examples of good practice such as the OPW's commitment to a multi-sectoral approach to flood risk management (OPW, 2018).
- 4. Universal clarity on terminology, fundamental concepts and appropriate risk data in the Irish context across all sectors to ensure coherence and harmonisation in actions and policies.
- 5. Identification of the most vulnerable groups and communities whose interests and needs should be prioritized to support their resilience to future shocks and participation in planning and decision-making.

6. Strengthening of local and national coordination and information-sharing mechanisms for the horizontal integration of DRM and CCA at national and sub-national level.

As highlighted in UNDRR's Global Assessment Report in 2019 – and more recently with the COVID-19 pandemic – non-linear change in climate is a novel reality, and threatens all three pillars of sustainable development, disaster management and climate action, increasing the propensity of system reverberations (UNDRR, 2019; 4). When certain communities, such as the DRM and CCA sectors, have an opportunity to align in their practices, that interconnectedness and collaboration will create synergistic responses and solutions for both, as well as the potential for cost-effective utilisation of resources in the medium- to long-term. The tighter knit they are, the less room there is for knowledge, information – and most importantly – *people*, to slip through the cracks. A harmonised approach, with all key stakeholders in concordance with one another, would make it easier to cover all bases required for integration. With an integrated CCA and DRM framework that centres increasing resilience and reducing vulnerability in Ireland, the outcomes can only be positive.

#### 5.1 The Next Steps

As mentioned in the project outline, the present review has been drafted as an inauguration in setting the scene for the future project deliverables in enhancing integration of climate change adaptation and disaster risk in Irish emergency planning. The following two deliverables – informed by key findings in the literature review – will not venture to identify specific technical solutions, but instead, determine successful practices and potential barriers for integration, by drawing on various strategies such as the ESPREssO and PLACARD Projects, and more significantly, the three case studies, to assist as benchmarks for successful integration at a national level. These case studies, informed by the stakeholder mapping in Work Package 2, will provide an opportunity for communities and key stakeholders to meaningfully contribute to the research by providing examples of good practices and challenges faced in the integration of CCA and DRM in Irish emergency planning, ideally leading to potential strengthening of resilience capacities and reduction of vulnerabilities.

# **References**<sup>2</sup>

- Banwell, N., Rutherford, S., Mackey, B., Chu, C., (2018), Towards improved linkage of disaster risk reduction and climate change adaptation in health: A review. *International Journal* of Environmental Research and Public Health, 15(4), p.793
- 2. Barrot J., Bharwani, S., (2017), *Visualising a harmonised language for CCA & DRR*, <<u>https://www.placard-network.eu/visualising-a-harmonised-language-for-cca-drr/></u>
- Bojovic, D., Giupponi, C., Karali, E., (2020), *How do CCA and DRR Actors Interact?*, PLACARD Interchange, <<u>https://www.placard-network.eu/how-do-cca-and-drr-actors-interact/</u>>
- 4. Cantwell, R., (2020), *Coastal Flooding and Climate Change in Ireland*, Gamma, Dublin, Ireland
- 5. Clarke, D., O'Donoghue Hynes, B., (2020), A Profile of Local Government Climate Actions in Ireland
- 6. Clarke, J., Acosta, E., (2019), *Determining the Costs of Extreme Flood Events to the Road Network in Ireland: An Example Application*
- Coninx, I., Suschenko, O., Michalek, G., Schwarze R., (2019), PLACARD, *Evolving Issues* Brief 2019
- Cubie, D., Natoli, T., 'Coherence, Alignment and Integration: Understanding the Legal Relationship Between Sustainable Development, Climate Change Adaptation and Disaster Risk Reduction' in Flood, S., LeTissier, M., Jerez Columbié, Y., O'Dwyer, B., *Creating Resilient Futures: Integrating Disaster Risk Reduction, Sustainable Development Goals and Climate Change Adaptation Agendas* (Palgrave Macmillan – forthcoming)
- 9. Department of Communications, Climate Action and the Environment (2017), *National Mitigation Plan Ireland*, Ireland
- Department of Communications, Climate Action and the Environment (2020), EU Emission Targets, <<u>https://www.dccae.gov.ie/en-ie/climate-action/topics/eu-and-international-climate-action/2020-eu-targets/Pages/default.aspx</u>>
- 11. Department of Communications, Climate Action and the Environment, (2018), *National Adaptation Framework: Planning for a climate resilient Ireland*, Dublin, Ireland
- 12. Department of Communications, Climate Action and the Environment, (2019), *Heads of Climate Action (Amendment) Bill 2019*, Dublin, Ireland

<sup>&</sup>lt;sup>2</sup> All internet links last accessed on 1<sup>st</sup> July 2020.

- 13. Department of Communications, Climate Action and the Environment, (2018) *Sectoral Planning Guidelines for Climate Change Adaptation*, Ireland
- 14. Department of Defence, (2017), A National Risk Assessment for Ireland, Ireland
- 15. Department of Defence, (2017), Government of Ireland, *Strategic Emergency Management National Structures and Framework*, Ireland
- Department of Housing, Planning and Local Government, (2006), Framework for Major Emergency Management, Ireland
- 17. Department of the Taoiseach, (2019), Government of Ireland, *National Risk Assessment-Overview of Strategic Risks*, Dublin, Ireland
- Desmond, M., O'Brien, P., McGovern, F., (2009), A Summary Of The State Of Knowledge On Climate Change Impacts For Ireland. Johnstown Castle Estate, Co. Wexford: Environmental Protection Agency
- 19. Diasa, N., Amaratunga, D., Haigha, R., (2017), *Challenges associated with integrating CCA and DRR in the UK*
- Djalante, R., Shaw, R., DeWit, A., (2020), Building resilience against biological hazards and pandemics: COVID-19 and its implications for the Sendai Framework. *Progress in Disaster Science*, p.100080
- 21. Environmental Protection Agency (EPA), *Ireland's Environment: Agriculture*, <<u>https://www.epa.ie/media/epa\_agriculture\_v2.pdf</u>>
- 22. European Commission, (2016), Action Plan on the Sendai Framework for Disaster Risk Reduction 2015-2030: A disaster risk-informed approach for all EU policies, Brussels
- 23. European Commission, (2018), *Adaptation preparedness scoreboard: Draft country fiche for Ireland*, <<u>https://ec.europa.eu/clima/sites/clima/files/consultations/docs/0035/ie\_en.pdf</u>>
- European Commission, (2020), *Effort sharing: Member States' emission targets*, accessed 28 May 2020 <<u>https://ec.europa.eu/clima/policies/effort\_en></u>
- 25. European Commission, (2020), *EU Adaptation Strategy*, <<u>https://ec.europa.eu/clima/policies/adaptation/what\_en#tab-0-1</u>>
- 26. European Environment Agency, (2017), EEA Report No.15/2017, *Enhancing coherence of the knowledge base, policies and practices*, Luxemburg
- Forzieri, G., Cescatti, A., e Silva, F.B., Feyen, L., (2017), Increasing risk over time of weatherrelated hazards to the European population: a data-driven prognostic study. *The Lancet Planetary Health*, 1(5), pp.e200-e208
- 28. Gleeson, E., McGrath R., Treanor M. (2013), *Ireland's Climate: The Road Ahead*, Dublin: Met Éireann

- Gotham, K.F., Campanella, R., (2010), Toward a research agenda on transformative resilience: Challenges and opportunities for post-trauma urban ecosystems. *Critical Planning*, *17*(Summer), pp.9-23
- Government of Ireland, (2019), A Better World: Ireland's Policy for International Development, Dublin, Ireland
- 31. Government of Ireland, (2019), *Climate Action Plan to Tackle Climate Breakdown*, Dublin, Ireland
- 32. Government of Ireland, (2015), *Climate Action and Low Carbon Development Act*, No.46/2015, Ireland
- 33. Government of Ireland, (2015), *Our Sustainable Future: A Framework for Sustainable Development in Ireland*, Ireland
- Government of Ireland, (2018), Project Ireland 2040 National Development Plan 2018-2027, Ireland
- 35. Hawchar, L., Ryan, P., Naughton, O., Nolan, P., (2018), *High-level Climate Change Vulnerability Assessment of Irish Critical Infrastructure*
- 36. Health Service Executive (HSE), (2020), Area-Major Emergency Plans, <a href="https://www.hse.ie/eng/services/list/3/emergencymanangement/area-mep/">https://www.hse.ie/eng/services/list/3/emergencymanangement/area-mep/</a>>
- 37. Houses of the Oireachtas, (2019), *Dáil Eireann*, <<u>https://www.oireachtas.ie/en/oireachtas-</u> tv/video-archive/dail-videos/2881>
- 38. Houses of the Oireachtas, (2019), *Report of the Joint Committee on Climate Action Climate Change: A Cross-Party Consensus for Action*, Dublin, Ireland
- Hugenbusch, D., Neumann, T., (2016), Cost-Benefit Analysis of Disaster Risk Reduction Executive Summary, Germany
- 40. International Federation of Red Cross and Red Crescent Societies, (2019), *The Cost of Doing Nothing: The Humanitarian Price of Climate Change and How it Can be Avoided*, Geneva
- IPCC, (2012), Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Mastrandrea, K.J. Mach, G.-K. Plattner, S.K. Allen, M. Tignor, and P.M. Midgley (eds.)]. Cambridge University Press, Cambridge, UK, and New York, NY, USA
- IPCC, (2014), Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Field, C.B., V.R. Barros, D.J. Dokken, K.J. Mach, M.D. Mastrandrea, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B.

Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L. White (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA

- 43. Ireland, (2018), Ireland: Voluntary National Review 2018 Report on the Implementation of the 2030 Agenda to the UN High-level Political Forum on Sustainable Development
- 44. ITF, (2016), Adapting Transport to Climate Change and Extreme Weather: Implications for Infrastructure Owners and Network Managers, ITF Research Reports, OECD Publishing, Paris
- 45. Burck, J., Hagen, U., Marten, F., Höhne, N., Bals, C., (2019) *Climate Change Performance Index*
- Leitner, M., Buschmann, D., Capela Lourenço, T., Coninx, I., Schmidt A., (2020), Bonding CCA and DRR: recommendations for strengthening institutional collaboration and capacities.
   PLACARD project, FC.ID: Lisbon
- 47. Leitner, M., Capela, T., Pulquério M., (2019), *Research and practice gaps, and their relevance* for Horizon Europe: outcomes from ECCA 2019, PLACARD project, Lisbon
- 48. Local Government Management Agency, (2020), *Climate Change Global Issue, Local Leadership*
- 49. McClean, D., (2020), *Will Resilience be the Buzzword for COVID 19 Recovery?*, UNDRR, <a href="https://www.undrr.org/news/will-resilience-be-buzz-word-covid-19-recovery">https://www.undrr.org/news/will-resilience-be-buzz-word-covid-19-recovery</a>
- 50. McClean, D. (2020), *Extreme Weather Events in a time of COVID-19*, UNDRR, <a href="https://www.undrr.org/news/extreme-weather-events-time-covid-19">https://www.undrr.org/news/extreme-weather-events-time-covid-19</a>>
- McDermott, R., Gibbons, P., (2015), Risk and Compliance with Normative Frameworks Relating to Disaster Management: Exploratory Case Studies from Indonesia and Ireland. Journal of International Humanitarian Legal Studies. 6. 345-376. 10.1163/18781527-00602005
- McMullan, C., Brown, G., O'Sullivan, D., (2016), Preparing to respond: Irish nurses' perceptions of preparedness for an influenza pandemic. International Emergency Nursing. 26. 3–7. 10.1016/j.ienj.2015.10.004
- 53. Mehryar, S., Surminski, S., (2020), The role of national laws in managing flood risk and increasing future flood resilience. Centre for Climate Change Economics and Policy Working Paper 365/Grantham Research Institute on Climate Change and the Environment Working Paper 334. London: London School of Economics and Political Science
- 54. Murphy, D., (2019), Alignment of country efforts under the 2030 Agenda, Paris Agreement and Sendai Framework, NAP Global Network, Uganda
- 55. NAP Global, (2020), Webinar: <u>Strengthening Alignment Between 'Adaptation to</u> <u>Climate Change – Disaster Risk Management – Sustainable Development'</u>

- 56. National Working Group for Major Emergency Management (2006), Multi-Agency Response to Flood Emergencies, <<u>http://mem.ie/wp-content/uploads/2016/11/Protocol-for-Multi-agency-response-to-Flood-Emergencies-18Nov-16.pdf</u>>
- 57. Natoli T., (2019), *Literature review on aligning climate change adaptation (CCA) and disaster risk reduction (DRR)*, IFRC | UCC, Geneva
- 58. Natoli T., (2020), Law and Policies that Protect the Most Vulnerable Against Climate-Related Disaster Risks: Findings and Lessons Learned from Pacific Island Countries, IFRC | UCC, Geneva
- Nolan, P., (2015), EPA Report: Ensemble of Regional Climate Model Projections for Ireland (No. 159), EPA climate change research report
- Ó Donnacha, S., (2015), Ireland: Statement made at the Third UN World Conference on Disaster Risk Reduction (WCDRR),
   <a href="https://www.preventionweb.net/english/professional/policies/v.php?id=44092">https://www.preventionweb.net/english/professional/policies/v.php?id=44092</a>>
- OECD, (2020), OECD Development Co-operation Peer Reviews: Ireland 2020, OECD Development Co-operation Peer Reviews, OECD Publishing, Paris, <a href="https://doi.org/10.1787/c20f6995-en">https://doi.org/10.1787/c20f6995-en</a>>
- 62. Office of Public Works, (2019), Government of Ireland, *Flood Risk Management Climate Change Adaptation Sectoral Plan*, Ireland
- 63. Office of Public Works, (2009), *The Planning System and Flood Risk Management, Guidelines* for Planning Authorities, Ireland
- 64. Office of Public Works, (2018), Public Consultation on the Draft of Risk Management Plans, <<u>http://s3-eu-west-</u> <u>1.amazonaws.com/docs.floodinfo.opw/floodinfo\_docs/Public\_Consultation\_Summary\_Report.</u> <u>pdf</u>>
- 65. Organisation for Economic Co-operation and Development, (2018), Policy Coherence for Sustainable Development Country Profiles <a href="https://www.oecd.org/governance/pcsd/Country%20Profile%20Ireland.pdf">https://www.oecd.org/governance/pcsd/Country%20Profile%20Ireland.pdf</a>>
- 66. O'Sullivan, K., (2018), Ireland will become leader on climate change, claims Taoiseach, Irish Times, <<u>https://www.irishtimes.com/news/environment/ireland-will-become-leader-on-climatechange-claims-taoiseach-1.3538665</u>>
- 67. Pollner, J., Kryspin-Watson, J., Nieuwejaar, S., (2010), *Disaster risk management and climate change adaptation in Europe and central Asia*. Washington, DC: World Bank
- 68. Prevention Web, (2017), Vulnerability, <<u>https://www.preventionweb.net/risk/vulnerability</u>>

- Sandholz, S., et al, (2020), Costs and benefits of (in)coherence: Disaster Risk Reduction in the Post-2015-Agendas: Synthesis Report, Bonn: United Nations University – Institute for Environment and Human Security (UNU-EHS)
- 70. S.I., N. Nicholls, D. Easterling, C.M. Goodess, S. Kanae, J. Kossin, Y. Luo, J. Marengo, K. McInnes, M. Rahimi, M. Reichstein, A. Sorteberg, C. Vera, and X. Zhang, (2012), Changes in climate extremes and their impacts on the natural physical environment. In: *Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation* [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Mastrandrea, K.J. Mach, G.-K. Plattner, S.K. Allen, M. Tignor, and P.M. Midgley (eds.)]. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change (IPCC). Cambridge University Press, Cambridge, UK, and New York, NY, USA, pp. 109-230
- Shine T., (2018-CCRP-SS.12), *Climate Resilient Ireland*, Environmental Protection Agency, Wexford, Ireland.
- 72. Shreve, C.M., Kelman, I., (2014), Does mitigation save? Reviewing cost-benefit analyses of disaster risk reduction. International Journal of Disaster Risk Reduction, 10, pp.213-235
- Siders, A.R., (2016). Incoherent Resilience Towards a Common Language for Climate Change Adaptation, Disaster Risk Reduction, and Sustainable Development. 10.1163/9789004318816\_006
- Soldi. R, (2016), Implementation of the Sendai Framework at the EU level: The Take up of Resilience Measures by Local and Regional Authorities, Committee of the Regions, European Union
- Solecki, W., Leichenko, R., O'Brien, K., (2011), Climate change adaptation strategies and disaster risk reduction in cities: Connections, contentions, and synergies. Current Opinion in Environmental Sustainability. 3. 135-141. 10.1016/j.cosust.2011.03.001
- 76. Thomalla, F., Downing, T., Spanger-Siegfried, E., Han, G., Rockström, J., (2006), Reducing hazard vulnerability: towards a common approach between disaster risk reduction and climate adaptation. *Disasters*, 30(1), pp.39-48
- 77. Tierney, K., (2014), *The Social Roots of Risk: Producing Disasters, Promoting Resilience* (*High Reliability and Crisis Management*). Stanford Business Books
- 78. Torney, D., Dekker, S., (2020), *The Irish Climate Policy Evaluation Project*, Dublin City University
- 79. Transport Infrastructure Ireland, (2017), *Strategy for Adapting to Climate Change on Ireland's Light Rail and National Road Network*
- UN General Assembly, *Transforming Our World: the 2030 Agenda for Sustainable Development*, 21 October 2015, A/RES/70/1

- 81. UNDRR, (2019), Global Assessment Report on Disaster Risk Reduction, Geneva, Switzerland
- 82. UNDRR, (2019), *Opportunities to Integrate Disaster Risk Reduction and Climate Resilience into Sustainable Finance*, Brussels Belgium, UNDRR, Regional Office for Europe
- 83. UNDRR (2017), Terminology: Online Glossary, <<u>https://www.undrr.org/terminology</u>>
- 84. UNDRR, (2020), Sendai Monitor
- 85. UNDRR, (2019), Opportunities to Integrate Disaster Risk Reduction and Climate Resilience into Sustainable Finance
- 86. UNFCCC/SBSTA/2017/2, Human health and adaptation: understanding climate impacts on health and opportunities for action, Bonn, Germany
- 87. UNFCCC, (2013), Warsaw International Mechanism for Loss and Damage associated with climate change impacts, Warsaw, Poland
- UNISDR, (2015), Making Development Sustainable: The Future of Disaster Risk Management. Global Assessment Report on Disaster Risk Reduction. Geneva, Switzerland
- UNISDR, (2015), Sendai Framework for Disaster Risk Reduction 2015-2030. Geneva, Switzerland
- 90. UNISDR, (2017), Build Back Better in Recovery, Rehabilitation and Reconstruction
- United Nations, Framework Convention on Climate Change (2015) Adoption of the Paris Agreement, 21<sup>st</sup> Conference of the Parties, Paris: United Nations.
- 92. United Nations Development Programme (UNDP), (2006), *Multi-Stakeholder Engagement Processes*
- 93. United Nations, Hyogo Framework for Action 2005-2015: Building the Resilience of Nations and Communities to Disasters, 22 January 2005, A/CONF.206/6, <a href="https://www.refworld.org/docid/42b98a704.html">https://www.refworld.org/docid/42b98a704.html</a>>
- Zuccaro, G., Leone, M., (2018), Building Resilient Cities: A Simulation-Based Scenario Assessment Methodology for the Integration of DRR and CCA in a Multi-Scale Design Perspective. *Procedia Engineering*, 212, pp.871-878
- 95. Zuccaro, G., Leone, M., Martucci, C., Grandjean, G., Baills, A., Schueller, L., Thieken, A., Thiebes, B., Schwarze, R., Pickl, S., Booth, L., Scolobig, A., Fleming, K., Clegg, G., (2018), ESPREssO Vision Paper on future research strategies following the Sendai Framework for DRR 2015-2030. 10.13140/RG.2.2.32883.99365